Macroeconomic Dynamics, 26, 2022, 1338–1374. Printed in the United States of America. doi:10.1017/S1365100520000589

# GROWTH AND CONVERGENCE THROUGH TECHNOLOGICAL INTERDEPENDENCE

**Wei Jin** *Tianjin University* 

**YIXIAO ZHOU** Australian National University

This paper presents a multi-country version of the Ramsey growth model with cross-country technological interdependence. The results rationalize several stylized facts about growth and convergence. First, individual countries tend to converge toward country-specific balanced growth paths rather than steady-state equilibria. Second, an economy that accounts for a smaller share of the world technology distribution harnesses the "advantages of backwardness" to catch up at a faster speed. Third, countries grow at different rates during the phase of transitional dynamics. However, technological interdependence creates a force toward cross-country convergence in the growth rate and stability of world income distribution in the long run. Finally, cross-country differences in structural characteristics and initial conditions lead to divergences in the level of income per capita.

Keywords: Convergence, Endogenous Growth, Technological Interdependence, Advantages of Backwardness

### 1. INTRODUCTION

The highly globalized world economy shows an overarching trend toward cross-country technological interdependence, where individual countries are simultaneously inventing and adopting each others technologies [e.g., European Commission (2013)]. In a world economy with technological interdependence, technological progress and economic growth in an individual country is not only a direct outcome of the fundamentals of its own but also depends on the characteristics of all other countries. This observation suggests that a better approximation of this reality might need a framework in which world technology depends on the

We are grateful to the detailed comments and advice of William Barnett (the editor), Oded Galor (the coeditor), and anonymous referees. We also thank Angus Chu, James Madison, and participants at: 2018 Asian Meeting of the Econometric Society (Seoul), 2019 Asian Meeting of the Econometric Society (Xiamen), 2019 Econometric Society Australasian Meeting (Perth), 2019 China International Conference in Macroeconomics (Shenzhen), and various seminars for helpful comments. The usual disclaimer applies. Address correspondence to: Wei Jin, Ma Yinchu School of Economics, Tianjin University, No. 92 Weijin Road, Tianjin 300072, China. e-mail: wei,jin19@outlook.com. Phone: (+86) 13616513164.

© 2020 Cambridge University Press 1469-8056/20



Source: OECD (2014) and World Bank (2017).

**FIGURE 1.** (a) Annual growth rate of GDP per capita between 1960 and 2000 versus log GDP per capita in 1960 for core Organisation for Economic Co-operation and Development (OECD) countries; (b) The evolution of income per capita in major economies, 1820-2000.

innovation efforts of all technologically interconnected countries, rather than on a single technology frontier country. For the analysis of growth and convergence in an interdependent world economy, the model incorporating cross-country technological interdependence provides a more satisfactory framework than the one with unidirectional technology diffusion from a single world frontier country [e.g., Nelson and Phelps (1966), Parente and Prescott (1994), Chu et al. (2014), Stokey (2015)].

This paper contributes to an analysis of growth and convergence through the lens of cross-country technological interdependence.<sup>1</sup> In doing that, we aim to rationalize the following stylized facts as shown in Figure 1. First, income per capita tends to increase along country-specific endogenous growth paths rather than to converge toward a common growth path or steady state. Second, an economy that, initially, had a low level of income tends to grow faster, and the correlation between the growth rate and the initial level is negative-the so-called  $\beta$ -convergence [e.g., Baumol (1986), Barro and Sala-i-Martin (1992), Galor (1996), Caselli et al. (1996)]. Third, although countries grow at different rates during transitional periods, there are more limited differences in sustained growth rates over recent years. Potential forces might exist in the world economy to ensure a relatively similar growth rate. Fourth and finally, the disparity in country-specific characteristics and initial conditions (e.g., preferences, technologies, population) could lead to cross-country divergence in the income level-the phenomenon coined as club convergence [e.g., Galor (1996), Azariadis (1996), Quah (1997)].<sup>2</sup>

We present a multi-country version of the Ramsey growth model, incorporating cross-country technological interdependence. Countries are technologically interdependent in the sense that the technology of individual economies both contributes to, and benefits from, a world technology pool. Technology diffusion from the world technology pool is determined by all technologically interdependent countries, rather than by a single world technology frontier that grows autonomously. This explicit account of cross-country technological interdependence is the main departure from the existing literature on technology diffusion and growth [e.g., Parente and Prescott (1994); Barro and Sala-i-Martin (1997); Chu et al. (2014); Benhabib et al. (2014); Stokey (2015); Acemoglu et al. (2006)].<sup>3</sup>

We show that the technologically interdependent world equilibrium is neither simply the equilibrium of each country on its own nor the equilibrium generated by unidirectional technology diffusion from a single frontier country. With cross-country technological interdependence, individual countries tend to grow along their country-specific balanced growth path, rather than converge toward the world technology frontier. A country with a wider technology gap can catch up at a faster rate by taking advantage of a greater amount of untapped knowledge in the rest of the world (RoW)-the so-called "advantages of backwardness" [e.g., Gerschenkron (1962), Rosenberg (1982), Abramovitz (1986)]. During the phase of transitional dynamics, individual countries grow at different rates, but the differences in growth rates are narrowing over time and will converge in the long run. Technological interdependence generates a pulling force to ensure cross-country convergence in growth rates. In particular, the common growth rate for crosscountry convergence is endogenously determined by structural characteristics of all technologically interdependent countries rather than a single world frontier economy that grows autonomously. These results rationalize the stylized fact of convergence in growth rates as in Howitt (2000), Acemoglu and Ventura (2002), and Howitt and Mayer-Foulkes (2005).

The technologically interconnected world economy has a stable long-run distribution in technology and income that depends on the structural characteristics of all individual countries. An economy with greater indigenous innovation efficiency and knowledge absorption capacity tends to account for a larger share in the world distribution. Meanwhile, when other countries enhance the indigenous innovation efficiency and knowledge absorptive capacity, the country under consideration will end up with a smaller share in the world distribution. The long-run trend in the world income distribution depends only on fundamental characteristics, independently of initial conditions. This result is consistent with the insights offered in the seminal paper of Acemoglu and Ventura (2002).

The pulling force in the interconnected world economy, working through technological interdependence, creates a force toward convergence in the growth rate. However, cross-country differences in both fundamental characteristics and initial conditions lead to divergences in the level of technology and income. This result coincides with the evidence of  $\sigma$ -convergence: the level of income per capita diverges across countries over time [e.g., Bernard and Durlauf (1995, 1996), Quah (1996, 1997)]. While having an initial wider technology gap can generate a faster pace of catch-up ("advantage of backwardness"), an economy with inferior country-specific characteristics and initial conditions (e.g., lower indigenous innovation efficiency, weaker knowledge absorptive capacity, and lower initial technology levels) would end up with a lower level of technology and income per capita in the long run. This result provides an informed interpretation of the *club convergence*: income levels tend to converge to one another only among countries with similar structural characteristics and initial conditions but diverge between countries with quite different ones [e.g., Galor (1996), Azariadis (1996), Howitt (2000), Acemoglu and Ventura (2002), Howitt and Mayer-Foulkes (2005)].

**Related Literature.** This paper relates mostly to the influential studies on the distance to frontier, catching up, and convergence. Nelson and Phelps (1966) provide an early contribution of technology diffusion and growth where technology diffusion depends on human capital investment and the technology gap relative to the world technology frontier. Parente and Prescott (1994) propose a theory of economic development in which domestic barriers that retard the inflow of technology from the world frontier account for the income disparity across countries. Barro and Sala-i-Martin (1997) find that a follower grows relatively fast and tends to catch up to the leaders by imitation, which is cheaper than invention. As the pool of copiable technology decreases, the costs of imitation tend to rise, and the followers growth rate tends to fall. Hence, a pattern of conditional convergence emerges in this model of the diffusion of technology. Acemoglu et al. (2006) show that certain institutional arrangements and policies that might initially increase the growth and speed of convergence could then lead to slower growth and failure to converge to the world technology frontier. Benhabib et al. (2014) present a stylized model of distance to frontier, in which countries choose a portfolio of innovation and imitation that facilitates technology diffusion from the world frontier for domestic productivity improvement. Stokey (2015) shows that the economy might either keep path with the technology frontier (sustained growth) or converge to a minimal technology level that is independent of the world frontier (stagnation). Chu et al. (2014) develop a Schumpeterian growth model in which economic growth in the developing country is driven by both domestic innovation and transfers of foreign technologies from the world technology frontier. Acemoglu et al. (2017) analyze asymmetric growth and institutions in an interdependent world based on a model of technologically interconnected countries. Building on Schumpeterian growth models with technology transfers, Aghion and Howitt (1998), Howitt (2000), and Howitt and Mayer-Foulkes (2005) show that countries that converge to a common growth rate tend to have different productivity levels. These seminal works consider technology diffusion from a single world technology frontier that grows autonomously. In contrast, the focus of this paper is on cross-country technological interdependence. The world technology is determined by all interdependent countries rather than a single frontier one.<sup>4</sup>

This paper also connects with the literature on technology diffusion and adoption. Eaton and Kortum (1996, 1999, Eaton and Kortum (2001)) estimate

the direction and magnitude of international technology diffusion. Caselli and Coleman (2001) examine the determinants of computer technology adoption. The linkages between productivity, technology adoption lags, and intensity of technology use are examined by, among others, Zeira (1998), Basu and Weil (1998), Acemoglu and Zilibotti (2001), Comin and Hobijn (2004, 2010), Schafer and Schneider (2015), Alesina et al. (2018), and Comin and Mestieri (2018). This strand of literature focuses on the process of technology diffusion and adoption, and the representations of linkages between technology diffusion and economic growth are parsimonious. In contrast, our analytical framework tends to be more disaggregated and provides richer specifications of the linkage between technology diffusion, growth, and convergence.

**Layout.** The rest of this paper is organized as follows. Section 2 presents the model. Section 3 characterizes the world equilibrium. Section 4 gives analytical results of growth and convergence. Section 5 provides calibrated simulations. Section 6 tests the theory empirically. Section 7 concludes.

### 2. THE MODEL

The framework builds on a multi-country version of the Ramsey growth model developed by Acemoglu (2009, Chapter 18), in which individual countries assimilate technology diffusion from a single world frontier country that grows autonomously at an exogenous rate. We differentiate our model by introducing cross-country technological interdependence, in which individual countries are technologically interdependent by simultaneously inventing and absorbing each other's technologies.<sup>5</sup> For a given country, technology diffusion from the RoW is determined by characteristics of all other countries, rather than by a single technology frontier country. Our explicit account of cross-country technological interdependence is the main departure from the existing literature which focus on unidirectional technology diffusions from a single world frontier country [e.g., Parente and Prescott (1994), Barro and Sala-i-Martin (1997), Acemoglu et al. (2006), Chu et al. (2014), Benhabib et al. (2014), Stokey (2015)].

**Demographics and Preference.** The world economy consists of *J* countries, indexed by j = 1, 2, ..., J. Each country admits a representative household. The number of populations within each representative household in country *j* at time *t* is given by  $L_j(t) = L_j(0) \exp(n_j t)$ , where  $n_j := \dot{L}_j/L_j$  is the rate (constant) of population growth and  $L_j(0) = 1$  the normalized initial condition.

The intertemporal utility is given by  $\int_0^\infty \exp(-\rho_j t)L_j(t)u(c_j(t))dt$ , where u(.) is the instantaneous utility function. The household in the economy j at time t has an asset holding of  $B_j(t)$  and the corresponding law of motion is given by  $\dot{B}_j(t) = r_j(t)B_j(t) + w_j(t)L_j(t) - c_j(t)L_j(t)$ , where  $r_j(t)$  is the rate of return on assets, and  $w_j(t)L_j(t)$  labor income earnings. Define by  $b_j(t) := B_j(t)/L_j(t)$  the per capita assets, the law of motion for the asset holding is rewritten as  $\dot{b}_j(t) = (r_j(t) - c_j(t) - c_j(t))$ .

 $n_j)b_j(t) + w_j(t) - c_j(t)$ . With a constant relative risk aversion utility function, the problem of the representative household in the economy *j* thus reads

$$\max_{\substack{[c_j(t),b_j(t)]_{t=0}^{\infty}}} \int_0^\infty \exp(-(\rho_j - n_j)t) \frac{c_j(t)^{1-\theta_j} - 1}{1 - \theta_j} dt \quad s.t.$$
$$\dot{b}_j(t) = (r_j(t) - n_j)b_j(t) + w_j(t) - c_j(t), \tag{1}$$

where  $c_j(t) := C_j(t)/L_j(t)$  is the consumption per capita,  $\rho_j$  is the discount rate, and  $\theta_j$  is the coefficient of relative risk aversion.

**Production and Capital Accumulation.** Each country admits a representative firm producing final goods with an aggregate production function as follows:

$$Y_j(t) = F(K_j(t), A_j(t)L_j(t)), \qquad (2)$$

where  $Y_j(t)$  and  $K_j(t)$  are the outputs and physical capital stocks in country *j* at time *t*, respectively. The households supply labor  $L_j(t)$  inelastically. The labor-augmenting technology,  $A_j(t)$ , is a country-specific and time-varying endogenous variable. The production technology F(.,.) satisfies the standard neoclassical assumptions and exhibits constant returns to scale. We proceed by defining the income per capita:

$$y_j(t) := \frac{Y_j(t)}{L_j(t)} = \frac{A_j(t)L_j(t)}{L_j(t)} F\left(\frac{K_j(t)}{A_j(t)L_j(t)}, 1\right) = A_j(t)f(k_j(t)),$$
(3)

where  $k_j(t) := \frac{K_j(t)}{A_j(t)L_j(t)}$  is the effective capital-labor ratio, and  $f(k_j(t)) := F\left(\frac{K_j(t)}{A_j(t)L_j(t)}, 1\right)$  is the intensive form of the production function. Hence, the firm solves a problem of profit maximization as follows:

$$\max_{[K_j(t),L_j(t)]_{t=0}^{\infty}} F\left(K_j(t), A_j(t)L_j(t)\right) - R_j(t)K_j(t) - w_j(t)L_j(t),\tag{4}$$

where the firm optimally chooses capital stocks  $K_j(t)$  and labor  $L_j(t)$  to maximize profits given the rental rate of capital  $R_j(t)$ , the wage rate  $w_j(t)$ , and the labor-augmenting technology  $A_j(t)$  in country *j* at time *t*. Given the production technology in (2), the stock of physical capital in country *j* is accumulated as follows:

$$K_j(t) = F(K_j(t), A_j(t)L_j(t)) - C_j(t) - \delta_j K_j(t),$$
(5)

where  $\delta_i$  is the rate of capital depreciation.

**Technological Interdependence.** Following the seminal works of Comin and Hobijn (2010) and Comin and Mestieri (2018), we consider that technological progress is driven by both undertaking indigenous innovation and absorbing foreign technology diffusion.<sup>6</sup> Hence, the law of motion for the technology in

country *j* at time *t* is specified as follows:

$$\dot{A}_{j}(t) = \lambda_{j}A_{j}(t) + \sigma_{j}(A_{WTP}(t) - A_{j}(t)) = \lambda_{j}A_{j}(t) + \sigma_{j}\left(\sum_{i=1}^{J} A_{i}(t) - A_{j}(t)\right).$$
 (6)

Here, the effect of indigenous innovation depends on the efficiency parameter  $\lambda_j$  and the existing technology level  $A_j(t)$ . The higher the current level of technology, the faster the pace of technological progress—the "standing on the shoulders of predecessor" effect [e.g., Romer (1990), Jones (1995, 1999)]. Meanwhile, technology diffusion is determined by knowledge absorptive capacity,  $\sigma_j$ , and the untapped technology in the world technology pool (WTP),  $A_{WTP}(t) - A_j(t)$ . In a technologically interdependent world economy, all individual countries feature complementary patterns of innovation specialization and thus contribute to the world technology as given by  $A_{WTP}(t) = \sum_{i=1}^{J} A_i(t)$ .<sup>7</sup> This specification is the point of departure from the existing literature that focuses on unidirectional technology diffusions from a single world frontier country.

Each country benefits from technology diffusion by assimilating untapped technology from the RoW according to the technology gap,  $A_{WTP} - A_j$ . A larger technology gap creates a greater amount of diffusion from the external world that potentially favors domestic technological progress—the "advantage of backwardness" proposed by Gerschenkron (1962). However, the "advantage of backwardness" does not necessarily mean that a country with a larger technology gap can absorb all untapped knowledge from abroad and immediately catch up. In fact, only part of the technologies available in the RoW can be absorbed effectively according to the country-specific knowledge absorptive capacity  $\sigma_j$ . In other words, a weaker knowledge absorptive capacity becomes an inhibitor that slows the effective adoption of technology diffusion, even if a greater amount of untapped technology is accessible in the external world [e.g., Zeira (1998), Comin and Hobijn (2004), Comin and Mestieri (2018), Alesina et al. (2018)].  $\lambda_j$  and  $\sigma_j$  are country-specific, reflecting cross-country differences in structural characteristics.

**The World Equilibrium.** For the above-described growth model with crosscountry technological interdependence, the world equilibrium is defined by the allocations  $[c_j(t), b_j(t), K_j(t), L_j(t), A_j(t), R_j(t), w_j(t)]$  in country j = 1, 2, ..., J at time  $t \in [0, \infty)$ , such that

- i. The household chooses consumption and asset allocations  $\left\{ [c_j(t), b_j(t)]_{t=0}^{\infty} \right\}_{j=1}^{J}$  to maximize intertemporal utility subject to the budget constraint given by (1);
- ii. The firm chooses capital and labor  $\{[K_j(t), L_j(t)]_{t=0}^{\infty}\}_{j=1}^{J}$  to maximize profits given by (4);
- iii. The stock of capital  $\{[K_j(t)]_{t=0}^{\infty}\}_{j=1}^{J}$  is accumulated according to (5);
- iv. The labor-augmenting technology  $\{[A_j(t)]_{t=0}^{\infty}\}_{j=1}^{J}$  evolves according to (6);

v. The rental rate of capital and the wage rate  $\{[R_j(t), w_j(t)]_{t=0}^{\infty}\}_{j=1}^{J}$  are such that both capital and labor markets clear. The asset holding by the household is equal to the capital stock used by the firm, and the labor supplied by the household is equal to the labor demanded by the firm.

#### 3. CHARACTERIZATION OF THE WORLD EQUILIBRIUM

**Technology Gap.** For analytical tractability, we define the proportional technology gap of country *j* relative to the world technology by:

$$a_j(t) := \frac{A_j(t)}{A_{WTP}(t)},\tag{7}$$

where  $a_j(t)$  is the share of country *j* in the world technology distribution. The law of motion for the technology (6) is thus rewritten by:

$$\dot{a}_j(t) = \sigma_j - \left(\sigma_j - \lambda_j + g_{WTP}(t)\right) a_j(t), \tag{8}$$

where the rates of technological progress specific to the world and an individual country are given, respectively, by:<sup>8</sup>

$$g_{WTP}(t) \coloneqq \frac{\dot{A}_{WTP}(t)}{A_{WTP}(t)} = \sum_{j=1}^{J} \left( \left( \lambda_j - \sigma_j \right) a_j(t) + \sigma_j \right),$$
$$g_j(t) \coloneqq \frac{\dot{A}_j(t)}{A_j(t)} = \lambda_j + \sigma_j \left( \frac{1}{a_j(t)} - 1 \right).$$
(9)

Combining (8) with (9), the law of motion for the proportional technology gap is determined by:

$$\dot{a}_j(t) = \sigma_j - \left(\sigma_j - \lambda_j + \sum_{j=1}^J \left( \left(\lambda_j - \sigma_j\right) a_j(t) + \sigma_j \right) \right) a_j(t).$$
(10)

**Effective Capital–labor Ratio.** Define consumption per effective unit of labor in country *j* at time *t* as  $\bar{c}_j(t) := \frac{C_j(t)}{A_j(t)L_j(t)}$ . The law of motion for the capital stock given in (5) is thus rewritten by  $\dot{k}_j(t) = f(k_j(t)) - \bar{c}_j(t) - (n_j + \delta_j + g_j(t)) k_j(t)$ , where  $g_j := \dot{A}_j/A_j$  is the rate of technological progress,  $n_j := \dot{L}_j/L_j$  the rate of population growth, and  $\delta_j$  the rate of capital depreciation. Substituting the second expression of (9) to replace  $g_j(t)$ , we derive the law of motion for the effective capital–labor ratio as follows:

$$\dot{k}_j(t) = f(k_j(t)) - \bar{c}_j(t) - \left(n_j + \delta_j + \lambda_j + \sigma_j\left(\frac{1}{a_j(t)} - 1\right)\right) k_j(t).$$
(11)

**Euler Equation of Consumption.** Solving the problem given by (1) yields the Euler equation:  $\dot{c}_j(t)/c_j(t) = \theta_j^{-1} (r_j(t) - \rho_j)$ , and the transversality condition:  $\lim_{t\to\infty} \exp(-(\rho_j - n_j)t)\mu_j(t)b_j(t) = 0$ . Meanwhile, solving the problem given by (4) determines the rental rate of capital and the wage rate as:  $R_j(t) = f'(k_j(t))$ ,

and  $w_j(t)/A_j(t) = f(k_j(t)) - k_j(t)f'(k_j(t))$ . The equilibrium rate of return on capital assets is given by  $r_j(t) = R_j(t) - \delta_j = f'(k_j(t)) - \delta_j$ . The Euler equation of consumption can thus boils down to:

$$\frac{\dot{\bar{c}}_j(t)}{\bar{c}_j(t)} = \frac{\dot{c}_j(t)}{c_j(t)} - \frac{\dot{A}_j(t)}{A_j(t)} = \frac{1}{\theta_j} \left( f'(k_j(t)) - \delta_j - \rho_j \right) - \lambda_j - \sigma_j \left( \frac{1}{a_j(t)} - 1 \right), \quad (12)$$

where  $\bar{c}_j(t) = \frac{c_j(t)}{A_j(t)} = \frac{C_j(t)}{A_j(t)L_j(t)}$  is consumption per effective unit of labor. The growth rate depends on the marginal product of capital  $f'(k_j(t))$ , and the rate of technological progress  $\dot{A}_j(t)/A_j(t)$  is given by (9).

**Transitional Dynamics.** The world equilibrium is characterized by the time path of the technology gap, effective capital–labor ratio, and consumption per effective unit of labor  $\{[a_j(t), k_j(t), \bar{c}_j(t)]_{t=0}^{\infty}\}_{j=1}^{J}$ . Starting from the initial condition  $[a_j(0), k_j(0), \bar{c}_j(0)]_{j=1}^{J}$ ,  $\{[a_j(t), k_j(t), \bar{c}_j(t)]_{t=0}^{\infty}\}_{j=1}^{J}$  evolve according to the equations (10)–(12) and converge toward the corresponding steady state  $[a_j^*, k_j^*, \bar{c}_j^*]_{j=1}^{J}$ . The stationary conditions of (10)–(12) determine  $[a_j^*, k_j^*, \bar{c}_j^*]_{j=1}^{J}$  as follows:

$$a_{j}^{*} = \frac{\sigma_{j}}{\sigma_{j} - \lambda_{j} + g_{WTP}^{*}}, \qquad f'(k_{j}^{*}) = \delta_{j} + \rho_{j} + \theta_{j}g_{WTP}^{*},$$
  
$$\bar{c}_{j}^{*} = f(k_{j}^{*}) - \left(\delta_{j} + n_{j} + g_{WTP}^{*}\right)k_{j}^{*}, \qquad (13)$$

where the superscript "\*" corresponds to the long-run balanced growth path<sup>9</sup> and  $g_{WTP}^* := \dot{A}_{WTP}^* / A_{WTP}^*$  is the long-run trend rate of world technological progress endogenously determined by:

$$\sum_{j=1}^{J} \frac{\sigma_j}{\sigma_j - \lambda_j + g_{WTP}^*} = 1.$$
 (14)

Here,  $g_{WTP}^*$  depends on structural characteristics of all interdependent countries in the world economy.

For any given initial condition of technology  $[A_1(0), A_2(t), ..., A_J(0)]$ , if there are some countries, say country j, of which the initial value of the technology gap is not equal to the steady state, that is,  $\frac{A_j(0)}{\sum_{j=1}^J A_j(0)} \coloneqq a_j(0) \neq a_j^* = \frac{\sigma_j}{\sigma_j - \lambda_j + g_{WTP}^*}$ , then the technologically interdependent world equilibrium has transitional dynamics from the initial condition toward the long-run balanced growth. Furthermore, the transitional dynamics are saddle-path stable. Given the initial values of  $[a_j(0), k_j(0)]_{j=1}^J$ , there is a stable manifold that determines the initial value of  $[\bar{c}_j(0)]_{j=1}^J$ , such that from the initial condition  $[a_j(0), k_j(0), \bar{c}_j(0)]_{j=1}^J$ ,  $\{[a_j(t), k_j(t), \bar{c}_j(t)]_{t=0}^\infty\}_{j=1}^J$  evolve according to (10), (11), and (12) and converge toward the long-run steady state  $[a_j^*, k_j^*, \bar{c}_j^*]_{j=1}^J$  determined by (13) (Appendix A provides details).

#### 4. GROWTH AND CONVERGENCE

#### 4.1. Convergence in the Growth Rates

In the long-run world equilibrium, proportional technology gap, consumption per effective unit of labor, and effective capital–labor ratio tend to be constant  $[a_j(t), k_j(t), \bar{c}_j(t)] = [a_j^*, k_j^*, \bar{c}_j^*]$  as given by (13). Hence, we derive the following result.

**PROPOSITION 1.** In a technologically interdependent world equilibrium, there is cross-country convergence in the growth rates of technology, consumption per capita, and income per capita, that is,

$$\frac{\dot{y}_{j}^{*}}{y_{i}^{*}} = \frac{\dot{c}_{j}^{*}}{c_{i}^{*}} = \frac{\dot{A}_{j}^{*}}{A_{i}^{*}} = \frac{\dot{A}_{WTP}^{*}}{A_{WTP}^{*}} \coloneqq g_{WTP}^{*},$$
(15)

where  $y_j := Y_j/L_j = A_j f(k_j)$ ,  $c_j := C_j/L_j = A_j \bar{c}_j$ , and  $A_j$  are income per capita, consumption per capita, and technology, respectively. The common growth rate for cross-country convergence is the long-run trend rate of world technological progress  $g^*_{WTP}$  as determined by (14).

Proposition 1 shows that Ramsey growth with cross-country technological interdependence can generate a trend of convergence in growth rates. This result rationalizes the empirical fact that there are more limited differences in the longrun sustained growth rates, though countries tend to grow at different rates during the phase of transitional dynamics [e.g., Baumol (1986), Williamson (1996)]. This paper isolates technological interdependence as a mechanism for cross-country convergence. This novel perspective complements the existing studies explaining convergence through the lens of international trade or technology transfers from the world technology frontier [e.g., Howitt (2000), Acemoglu and Ventura (2002), Howitt and Mayer-Foulkes (2005)]. Furthermore, the growth rate for convergence is endogenous and depends on the structural characteristics of all individual countries in a technologically interdependent world economy. This result departs from the existing literature, which views each economy as converging to the growth rate of a single world frontier country that advances technology autonomously [e.g., Parente and Prescott (1994), Barro and Sala-i-Martin (1997), Acemoglu et al. (2006), Chu et al. (2014), Benhabib et al. (2014), Stokey (2015)].

In a world equilibrium with cross-country technological interdependence, the common growth rate for convergence depends on structural characteristics of all individual countries. The following result gives the comparative static effects of indigenous innovation efficiency and knowledge absorptive capacity.

COROLLARY 1. The comparative static effects of indigenous innovation efficiency  $\lambda_j$  and knowledge absorptive capacity  $\sigma_j$  on the common growth rate for cross-country convergence  $g^*_{WTP}$  are given, respectively, by

$$\frac{\partial g_{WTP}^*}{\partial \lambda_j} = \frac{\sigma_j}{\left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2 \sum_{i=1}^J \frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2}} > 0,$$
(16a)

$$\frac{\partial g_{WTP}^*}{\partial \sigma_j} = \frac{g_{WTP}^* - \lambda_j}{\left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2 \sum_{i=1}^J \frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2}} > 0.$$
(16b)

Proof. See Appendix B.

The interdependent world economy tends to grow at a larger rate if some national economies improve their indigenous innovation efficiencies and knowledge absorptive capacities. As technological interdependence drives crosscountry convergence to the rate of world technological progress, the improvement in structural characteristics of one country has a positive effect in promoting the long-run growth of other countries.

#### 4.2. World Technology Distribution

In a technologically interconnected world, changes in one country's structural characteristics affect another country's position in the world technology distribution. While technological interdependence ensures cross-country convergence in the rate of technological progress, there tends to be divergence in the world technology distribution due to cross-country disparity in fundamental characteristics. Specifically, following the first expression of (13), the long-run distribution of the world technology boils down to  $a_j^* = A_j^*/A_{WTP}^* = \sigma_j/(\sigma_j - \lambda_j + g_{WTP}^*)$ . The following result shows the effect of country characteristics on the world technology distribution.

**PROPOSITION 2.** The within- and cross-country effects of country characteristics on the world technology distribution are given, respectively, by:

$$\frac{\partial a_{j}^{*}}{\partial \lambda_{j}} = \left(\sum_{i \neq j} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}}\right) \frac{\partial g_{WTP}^{*}}{\partial \lambda_{j}}$$

$$= \frac{\left(\sum_{i \neq j} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}}\right) \sigma_{j}}{\left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2} \sum_{i=1}^{J} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}}} > 0, \quad (17a)$$

$$\frac{\partial a_{j}^{*}}{\partial \sigma_{j}} = \left(\sum_{i \neq j} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}}\right) \frac{\partial g_{WTP}^{*}}{\partial \sigma_{j}}$$

$$= \frac{\left(\sum_{i \neq j} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}}\right) \left(g_{WTP}^{*} - \lambda_{j}\right)}{\left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2} \sum_{i=1}^{J} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}} > 0, \quad (17b)$$

 $\sigma$ :

$$\frac{\partial a_i^*}{\partial \lambda_j} = -\frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2} \frac{\partial g_{WTP}^*}{\partial \lambda_j} = -\frac{\overline{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2} \sigma_j}{\left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2 \sum_{i=1}^J \frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2}} < 0,$$

$$\frac{\partial a_i^*}{\partial \sigma_i} = -\frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2} \frac{\partial g_{WTP}^*}{\partial \sigma_i} = -\frac{\overline{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2} \left(g_{WTP}^* - \lambda_j\right)}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2} < 0,$$

$$(17c)$$

$$\frac{\partial a_i}{\partial \sigma_j} = -\frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2} \frac{\sigma_{SWTP}}{\partial \sigma_j} = -\frac{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)}{\left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2 \sum_{i=1}^J \frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2}} < 0$$
(17d)

where  $\frac{\partial a_j^*}{\partial \lambda_j}$  and  $\frac{\partial a_j^*}{\partial \sigma_j}$  are the effect of the country j's indigenous innovation efficiency and knowledge absorptive capacity on this country's technology share.  $\frac{\partial a_i^*}{\partial \lambda_j}$  and  $\frac{\partial a_i^*}{\partial \sigma_j}$  are the country j's indigenous innovation efficiency and knowledge absorptive capacity on another country i's technology share.

#### Proof. See Appendix C.

Proposition 2 emphasizes that we should often consider the world economy as an interconnected equilibrium in which one country's technological position depends on characteristics of all other countries, not simply an equilibrium of each country on its own. While technological interdependence creates a force driving convergence in the growth rates, the shares that different countries account for in the world technology distribution tend to diverge due to cross-country disparity in fundamental characteristics. This result coincides with the insights offered by the seminal works of Howitt (2000), Acemoglu and Ventura (2002), and Howitt and Mayer-Foulkes (2005). More specifically, an economy with a stronger capacity to undertake indigenous innovation and absorb technology diffusion will end up with a larger share in the world technology distribution. Furthermore, changes in a particular country's characteristics have effects on another country's technological position. The negative cross-country effect suggests that improvement in one country's indigenous innovation efficiencies and knowledge absorptive capacity will shrink the share of other countries in the world technology distribution.

#### 4.3. Divergences in Levels

The pulling force in the interconnected world economy, working through technological interdependence, drives technological progress and ensures cross-country convergence in the long-run growth rates. However, we will show below that there tends to be cross-country divergence in the aggregate level, owing to cross-country disparity in fundamental structural characteristics. For analytical tractability, we use the relative ratio between countries (proportional level differences) as a proxy for cross-country divergence and obtain the following proposition. **PROPOSITION 3.** The technology ratio between any given two countries i, j = 1, 2, ..., J is determined by:

$$\frac{A_i^*}{A_j^*} = \frac{\frac{\sigma_i}{(\sigma_i - \lambda_i + g_{WTP}^*)}}{\frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)}},$$
(18)

the capital ratio between countries is determined by:

$$\frac{K_i^*}{K_j^*} = \frac{\frac{\sigma_i}{(\sigma_i - \lambda_i + g_{WTP}^*)}}{\frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)}} \cdot \frac{f'^{-1}\left(\delta_i + \rho_i + \theta_i g_{WTP}^*\right)}{f'^{-1}\left(\delta_j + \rho_j + \theta_j g_{WTP}^*\right)} \cdot \frac{L_i^*}{L_j^*},\tag{19}$$

and the consumption ratio between countries is determined by:

$$\frac{C_{i}^{*}}{C_{j}^{*}} = \frac{\frac{\sigma_{i}}{(\sigma_{i}-\lambda_{i}+g_{WTP}^{*})}}{\frac{\sigma_{j}}{(\sigma_{j}-\lambda_{j}+g_{WTP}^{*})}} \cdot \frac{f(f'^{-1}\left(\delta_{i}+\rho_{i}+\theta_{i}g_{WTP}^{*}\right)) - \left(\delta_{i}+n_{i}+g_{WTP}^{*}\right)f'^{-1}\left(\delta_{i}+\rho_{i}+\theta_{i}g_{WTP}^{*}\right)}{f(f'^{-1}\left(\delta_{j}+\rho_{j}+\theta_{j}g_{WTP}^{*}\right)) - \left(\delta_{j}+n_{j}+g_{WTP}^{*}\right)f'^{-1}\left(\delta_{j}+\rho_{j}+\theta_{j}g_{WTP}^{*}\right)} \cdot \frac{L_{i}^{*}}{L_{j}^{*}},$$
(20)

where  $f'^{-1}(\cdot)$  is the inverse function (denoted by "<sup>-1</sup>") of the derivative (denoted by "'") of the intensive-form production function  $f(k) := F\left(\frac{K}{AL}, 1\right)$ .  $[\lambda, \sigma, \delta, \theta, \rho]$ are indigenous innovation efficiency, knowledge absorptive capacity, capital depreciation rate, coefficient of relative risk aversion, and rate of time preference, respectively.  $g_{WTP}^*$  is the long-run trend rate of world technological progress determined by (14). The labor ratio between countries is given by  $L_i^*/L_j^* =$  $\exp((n_i - n_j)t^*)$ , where n is the growth rate of labor and  $t^*$  is the time point in the long-run balanced growth phase. Cross-country differences in structural characteristics  $[\lambda, \sigma, \delta, \theta, \rho, n]$  lead to cross-country divergence in the level of technology, capital stocks, and consumption, and the divergence becomes wider as time proceeds  $t^* \to \infty$ .

Proof. See Appendix D.

Proposition 3 shows that the aggregate levels of technology, capital stocks, and consumption tend to diverge, owing to cross-country disparity in structural characteristics such as indigenous innovation efficiency and knowledge absorptive capacity. The following result gives the comparative static effects.

**PROPOSITION 4.** The cross-country divergence in technology, capital stocks, and consumption, measured by  $A_i^*/A_j^*$ ,  $K_i^*/K_j^*$  and  $C_i^*/C_j^*$ , respectively, depends on indigenous innovation efficiency and knowledge absorptive capacity  $[\lambda_i, \sigma_i, \lambda_j, \sigma_j]$ .

*i.* The comparative static effects of  $[\lambda_i, \sigma_i, \lambda_j, \sigma_j]$  on divergence in technology are given by:

$$\frac{\partial (A_i^*/A_j^*)}{\partial \lambda_i} > 0, \quad \frac{\partial (A_i^*/A_j^*)}{\partial \sigma_i} > 0, \quad \frac{\partial (A_i^*/A_j^*)}{\partial \lambda_j} < 0, \quad \frac{\partial (A_i^*/A_j^*)}{\partial \sigma_j} < 0.$$
(21)

ii. If the following condition holds

$$\frac{\theta_{i}}{k_{i}^{*}f''(k_{i}^{*})} - \frac{\theta_{j}}{k_{j}^{*}f''(k_{j}^{*})} + \frac{1}{a_{i}}\sum_{j\neq i}\frac{\sigma_{j}}{\left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2}} - \frac{1}{a_{j}}\frac{\sigma_{j}}{\left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2}} > 0,$$
(22)

the comparative static effects of  $[\lambda_i, \sigma_i, \lambda_j, \sigma_j]$  on divergence in capital stocks are given by:

$$\frac{\partial(K_i^*/K_j^*)}{\partial\lambda_i} > 0, \quad \frac{\partial(K_i^*/K_j^*)}{\partial\sigma_i} > 0, \quad \frac{\partial(K_i^*/K_j^*)}{\partial\lambda_j} < 0, \quad \frac{\partial(K_i^*/K_j^*)}{\partial\sigma_j} < 0.$$
(23)

Otherwise the comparative static effects are opposite, that is,

$$\frac{\partial(K_i^*/K_j^*)}{\partial\lambda_i} < 0, \quad \frac{\partial(K_i^*/K_j^*)}{\partial\sigma_i} < 0, \quad \frac{\partial(K_i^*/K_j^*)}{\partial\lambda_j} > 0, \quad \frac{\partial(K_i^*/K_j^*)}{\partial\sigma_j} > 0.$$
(24)

iii. If the following condition holds

$$\frac{1}{\bar{c}_{i}^{*}}\left(\left((\theta_{i}-1)g_{WTP}^{*}+\rho_{i}-n_{i}\right)\frac{\theta_{i}}{f''(k_{i}^{*})}-k_{i}^{*}\right) - \frac{1}{\bar{c}_{j}^{*}}\left(\left((\theta_{j}-1)g_{WTP}^{*}+\rho_{j}-n_{j}\right)\frac{\theta_{j}}{f''(k_{j}^{*})}-k_{j}^{*}\right) + \frac{1}{a_{i}^{*}}\sum_{j\neq i}\frac{\sigma_{j}}{\left(\sigma_{j}-\lambda_{j}+g_{WTP}^{*}\right)^{2}}-\frac{1}{a_{j}^{*}}\frac{\sigma_{j}}{\left(\sigma_{j}-\lambda_{j}+g_{WTP}^{*}\right)^{2}}>0, \quad (25)$$

the comparative static effects of  $[\lambda_i, \sigma_i, \lambda_j, \sigma_j]$  on divergence in consumption are given by:

$$\frac{\partial(C_i^*/C_j^*)}{\partial\lambda_i} > 0, \quad \frac{\partial(C_i^*/C_j^*)}{\partial\sigma_i} > 0, \quad \frac{\partial(C_i^*/C_j^*)}{\partial\lambda_j} < 0, \quad \frac{\partial(C_i^*/C_j^*)}{\partial\sigma_j} < 0.$$
(26)

Otherwise the comparative static effects are opposite, that is,

$$\frac{\partial (C_i^*/C_j^*)}{\partial \lambda_i} < 0, \quad \frac{\partial (C_i^*/C_j^*)}{\partial \sigma_i} < 0, \quad \frac{\partial (C_i^*/C_j^*)}{\partial \lambda_j} > 0, \quad \frac{\partial (C_i^*/C_j^*)}{\partial \sigma_j} > 0.$$
(27)

Proof. See Appendix E.

Proposition 4 shows that an economy with higher indigenous innovation efficiencies and knowledge absorption capacities tends to have a higher technology level as compared to other countries. However, increases in one country's indigenous innovation efficiencies and knowledge absorption capacities might not necessarily lead to a larger stock of capital and a higher level of consumption as compared to other countries. As Proposition 4 shows, when conditions (22) and (25) are met, the improvement of one country's indigenous innovation efficiencies and knowledge absorption capacities can generate a larger stock of capital and a higher level of consumption relative to other countries.

#### 1352 WEI JIN AND YIXIAO ZHOU

The intuition of (22) is as follows. The effect of indigenous innovation efficiency  $\lambda_i$  on country *i* and *j*'s effective capital–labor ratio through the channel of world technological progress is characterized by  $\frac{1}{k_i^*} \frac{\partial k_i^*}{\partial \lambda_i} = \frac{\theta_i}{k_i^* f''(k_i^*)} \frac{\partial g_{WTP}^*}{\partial \lambda_i}$  and  $\frac{1}{k_i^*} \frac{\partial k_j^*}{\partial \lambda_i} = \frac{\theta_j}{k_i^* f''(k_i^*)} \frac{\partial g_{WTP}^*}{\partial \lambda_i}, \text{ respectively. If } \frac{\theta_i}{k_i^* f''(k_i^*)} - \frac{\theta_j}{k_i^* f''(k_i^*)} > 0, \text{ an increase in } \lambda_i$ generates a positive effect to widen cross-country differences (between country *i* and *j*) in the effective capital–labor ratio, that is,  $\frac{\partial (k_i^*/k_j^*)}{\partial \lambda_i} = \frac{k_i^*}{k_i^*} (\frac{\theta_i}{k_i^* f''(k_i^*)})$  $\frac{\theta_j}{k_i^* f''(k_j^*)} \frac{\partial g_{WTP}^*}{\partial \lambda_i} > 0$ . Meanwhile, the effect of indigenous innovation efficiency  $\lambda_i$ on country *i* and *j*'s technology gap through the channel of world technologi-cal progress is characterized by  $\frac{1}{a_i^*} \frac{\partial a_i^*}{\partial \lambda_i} = (\frac{1}{a_i^*} \sum_{j \neq i} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2}) \frac{\partial g_{WTP}^*}{\partial \lambda_i}$  and  $\frac{1}{a_j^*} \frac{\partial a_j^*}{\partial \lambda_i} =$  $\frac{1}{a_j^*} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} \frac{\partial g_{WTP}^*}{\partial \lambda_i}, \text{ respectively. If } \frac{1}{a_i} \sum_{j \neq i} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} - \frac{1}{a_j} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} > 0,$ an increase in  $\lambda_i$  will widen the cross-country differences in the technology gap, that is,  $\frac{\partial (a_i^*/a_j^*)}{\partial \lambda_i} = \frac{a_i^*}{a_j^*} (\frac{1}{a_i} \sum_{j \neq i} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} - \frac{1}{a_j} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2}) \frac{\partial g_{WTP}^*}{\partial \lambda_i} > 0.$  Combining the effects on both effective capital-labor ratio and technology gap, we can accordingly show an increase in  $\lambda_i$  will widen the cross-country difference in capital stock if the sufficient condition (22) is met, because we have  $\frac{\partial (K_i^*/K_j^*)}{\partial \lambda_i} =$  $\left(\frac{a_i^*}{a_j^*} \cdot \frac{\partial(k_i^*/k_j^*)}{\partial \lambda_i} + \frac{k_i^*}{k_j^*} \cdot \frac{\partial(a_i^*/a_j^*)}{\partial \lambda_i}\right) \frac{L_i^*}{L_i^*}.$  A similar reasoning applies to the sufficient condition (25) ensuring the comparative static effects on cross-country divergence in consumption given in (26).

#### 4.4. Advantage of Backwardness

The analysis proceeds by understanding cross-country growth patterns from the perspective of "advantage of backwardness," that is, a technologically backward economy tends to catch up at a faster pace [e.g., Gerschenkron (1962)]. Specifically, the technology position of a given country j = 1, 2, ..., J in the world distribution is measured by the proportional technology gap  $a_j(t) = \frac{A_j(t)}{\sum_{j=1}^J A_j(t)}$ , and transition dynamics along the stable saddle path is determined by:

$$a_j(t) = a_j(0) \exp(-|\chi_j|t) + a_j^* (1 - \exp(-|\chi_j|t)),$$
(28)

where the technology gap starts from the initial condition  $a_j(0)$  and converges toward the long-run steady state  $a_j^*$ , and the speed of transition dynamics is governed by the negative stable eigenvalue  $\chi_j$  (Appendix A provides details):

$$\chi_j = -g_{WTP}^* - (\lambda_j - \sigma_j)(a_j^* - 1) = -\left(\frac{\sigma_j}{a_j^*} + (\lambda_j - \sigma_j)a_j^*\right) < 0.$$
(29)

We thus establish the following proposition to explain the advantage of economic backwardness.

**PROPOSITION 5.** A technologically backward country *j* with a smaller share in the world technology distribution,  $a_j^* := A_j^* / \sum_{j=1}^J A_j^*$ , takes the "advantage of backwardness" to converge toward its corresponding balanced growth path at a faster speed.

Proof. See Appendix F.

A country that is relatively "backward" tends to make faster transitions to its corresponding long-run balanced growth path, and this is because technologically backward economies, with access to a larger amount of untapped knowledge in the RoW, can assimilate foreign technology diffusion to advance their domestic technology level. The pulling force attributable to the Gerschenkron's "advantage of backwardness" tends to be stronger to facilitate catching up at a faster pace [e.g., Gerschenkron (1962)]. This result can shed light on the stylized fact of  $\beta$ -convergence: an economy that initially had a low level of income tends to grow faster, and the rate of income growth over time and the initial income level are correlated negatively [e.g., Baumol (1986), Barro and Sala-i-Martin (1992), Galor (1996), Caselli et al. (1996), Barro (2015)]. Note that, while technologically backward economies can harness the "advantage of backwardness" to experience a fast pace of transitional dynamics, their long-run growth trend will still converge to the growth rate that is common to all interdependent countries. Also, technologically backward economies will end up with a lower position in the world technology distribution, owing to their lower indigenous innovation efficiency and knowledge absorptive capacity.

#### 5. QUANTITATIVE ANALYSIS

### 5.1. Model Calibration

Parameterization. The quantitative analysis considers the world economy as three countries/regions: the USA, the EU, and the RoW. The model is calibrated using the Penn World Table (PWT) version 9.1 database which covers 182 countries between 1950 and 2017 [Feenstra et al. (2015)]. The calibrated parameters are given in Table 1. The average growth rates of labor are estimated as  $n_{US} = 1.4\%$ ,  $n_{EU} = 0.5\%$ ,  $n_{RoW} = 2.0\%$ . The average rates of capital depreciation are set at  $\delta_{US} = 3.82\%$ ,  $\delta_{EU} = 3.56\%$ , and  $\delta_{RoW} = 3.36\%$  according to the PWT. The rate of time preference is set at a standard value  $\rho = 0.02$  [e.g., Grossmann et al. (2013), van der Ploeg and Withagen (2014)]. We use  $\theta = 2$  as an implied coefficient of relative risk aversion that is within the consensus range 1-3 [e.g., Mehra and Prescott (1985), Epstein and Zin (1991), Acemoglu et al. (2012)].<sup>10</sup> The production technology is specified as a Cobb–Douglas function,  $Y = F(K, AL) = K^{\beta}(AL)^{1-\beta}$ . The corresponding intensive-form representation is  $F\left(\frac{K}{AL},1\right) := f(k) = k^{\beta}$ , where the factor share of labor income is given by  $1 - \beta$ . We use the PWT database to calculate the factor share of labor compensation in GDP, and the average values over the period 1950–2017 are estimated as 0.62,

Description	Parameter	Value		
		j=US	j=EU	j=RoW
Indigenous innovation efficiency	$\lambda_i$	0.03	0.025	0.02
Knowledge absorptive capacity	$\sigma_i$	0.02	0.015	0.01
Output elasticity of capital	$\dot{\beta_i}$	0.38	0.39	0.4
Growth rates of labor	$n_i$	0.014	0.005	0.02
Rate of capital depreciation	$\delta_i$	0.038	0.036	0.034
Coefficient of relative risk aversion	$\theta_i$	2	2	2
Rate of time preference	$\rho_i$	0.02	0.02	0.02
Initial values of technology gap	$a_i(0)$	0.289	0.385	0.326
Initial values of effective capital-labor ratio	$k_j(0)$	22.98	11.22	2.65

### TABLE 1. Calibrated parameters for quantitative analysis

0.61, and 0.6 for the USA, the EU, and RoW, respectively. Hence, the output elasticities of capital (the factor share of capital) are set at  $\beta_{US} = 0.38$ ,  $\beta_{EU} = 0.39$  and  $\beta_{RoW} = 0.4$ .

We calibrate the parameters of indigenous innovation efficiency and knowledge absorptive capacity as follows. We collect the date of intangible capital stocks from the PWT database as a measurement of the technology level. The PWT 9.1 documents the data of intangible capital stocks over the period 1950– 2017.<sup>11</sup> As Figure 2(a)–(c) shows, we fit the trend of intangible capital stocks for the USA, the EU, and RoW using the law of motion for the technology given by (6) with the calibrated parameters  $\lambda_{US} = 0.03$ ,  $\lambda_{EU} = 0.025$ ,  $\lambda_{RoW} =$ 0.02,  $\sigma_{US} = 0.02$ ,  $\sigma_{EU} = 0.015$ , and  $\sigma_{RoW} = 0.01$ . Aggregating the three countries/regions gives the calibration for the world as shown in Figure 2(d). With the parameters  $[\lambda_j, \sigma_j]_{j=US,EU,RoW}$ , the long-run trend rate of world technological progress is endogenously determined by:

$$\sum_{j=US,EU,RoW} \frac{\sigma_j}{\sigma_j - \lambda_j + g^*_{WTP}} = 1 \quad \Rightarrow \quad g^*_{WTP} = 0.055.$$

The calibrated parameters also satisfy the condition  $\sigma_j < \lambda_j < g^*_{WTP}$ , given that the technology gap is less than unity,  $\sigma_j/(\sigma_j - \lambda_j + g^*_{WTP}) < 1 \Rightarrow \lambda_j < g^*_{WTP}$ . Undertaking indigenous innovation contributes more to domestic technological progress than absorbing foreign technology diffusion (no free riding behavior).

**Boundary Conditions.** We use the PWT data of intangible capital stocks for the year 1950 (14,341, 19,130, and 16,199 in millions 2011 USD for the USA, the EU, and RoW, respectively) to estimate the initial condition of technology gaps as  $[a_{US}(0), a_{EU}(0), a_{RoW}(0)] = [0.289, 0.385, 0.326]$ . With the PWT data of physical capital  $K_j(0)$  and labor  $L_j(0)$ , we use the formula  $k_j(0) = K_j(0)/(A_j(0)L_j(0))$  to calculate the initial condition of the effective capital–labor ratio as  $[k_{US}(0), k_{EU}(0), k_{RoW}(0)] = [22.98, 11.22, 2.65]$ . Hence, the initial values of predetermined state variables  $[a_i(0), k_i(0)]_{i=US,EU,RoW}$  give the initial condition



**FIGURE 2.** Panel (a) fits the US data of intangible capital stocks using the law of motion for the technology (6) with the calibrated parameters  $\lambda_{US} = 0.03$  and  $\sigma_{US} = 0.02$ . Panel (b) fits the EU data of intangible capital stocks using the law of motion for the technology (6) with the calibrated parameters  $\lambda_{EU} = 0.025$  and  $\sigma_{US} = 0.015$ . Panel (c) fits the RoW data of intangible capital stocks using the law of motion for the technology (6) with the calibrated parameters  $\lambda_{RoW} = 0.02$  and  $\sigma_{RoW} = 0.01$ . Panel (d) gives the calibration for the world technology by aggregating the USA, the EU, and RoW.

of differential equations (10)–(12). Meanwhile, we use (13) to compute the longrun Balanced Growth Path (BGP) and establish the terminal condition as follows:

$$a_{j}^{*} = \frac{\sigma_{j}}{\sigma_{j} - \lambda_{j} + g_{WTP}^{*}}, \qquad k_{j}^{*} = \left(\frac{\delta_{j} + \rho_{j} + \theta_{j} g_{WTP}^{*}}{\beta_{j}}\right)^{\frac{1}{\beta_{j}-1}},$$
$$\bar{c}_{j}^{*} = \left(k_{j}^{*}\right)^{\beta_{j}} - \left(\delta_{j} + n_{j} + g_{WTP}^{*}\right)k_{j}^{*}.$$

This boils down to  $[a_{US}^*, a_{EU}^*, a_{RoW}^*] = [0.44, 0.33, 0.23], [k_{US}^*, k_{EU}^*, k_{RoW}^*] = [3.72, 4.07, 4.44], and [\bar{c}_{US}^*, \bar{c}_{EU}^*, \bar{c}_{RoW}^*] = [1.25, 1.34, 1.33].$ 

**Solving Transitional Dynamics.** With the initial and terminal conditions, we solve the differential equations numerically with the boundary value problem characterized by (10)–(12). As Figure 3 shows, there is a stable saddle path that determines the initial value of  $[c_j(0)]_{j=US,EU,RoW}$  endogenously, given the initial



**FIGURE 3.** Panel(a) depicts transitional dynamics of technology gap  $a_j$ , effective capital– labor ratio  $k_j$ , and consumption per effective unit of labor  $\bar{c}_j$ . Panel (b) plots the time path of technology gap  $a_j(t) = A_j(t)/A_{WTP}(t)$ . Panel (c) plots the time path of effective capital–labor ratio  $k_j(t) = K_j(t)/(A_j(t)L_j(t))$ . Panel (d) plots the time path of consumption per effective unit of labor  $\bar{c}_j(t) = C_j(t)/(A_j(t)L_j(t))$ . The dashed red, solid blue, and dash-dotted black lines correspond to the USA, the EU, and RoW, respectively.

values of predetermined state variables  $[a_j(0), k_j(0)]_{j=US,EU,RoW}$ . Then, starting from the initial condition  $[a_j(0), k_j(0), \tilde{c}_j(0)]_{j=US,EU,RoW}$ , the world equilibrium evolves along the stable saddle path of transitional dynamics toward the long-run balanced growth phase  $[a_i^*, k_i^*, \tilde{c}_i^*]_{j=US,EU,RoW}$ .

#### 5.2. Simulation Results

**Convergence in Growth Rates.** Figure 4(a), (b), and (c) shows the cross-country pattern of convergence in the growth rates of technology, income, and consumption per capita, respectively. The USA, the EU, and RoW with different structural characteristics and initial conditions increase their technology and income levels at different rates during the phase of transitional growth. However, the difference is narrowing over time, and there tends to be cross-country convergence in



**FIGURE 4.** Panel (a), (b), and (c) plots cross-country convergence in the rate of technological progress  $\dot{A}_j/A_j$ , the growth rate of income per capita  $\dot{y}_j/y_j$ , and the growth rate of consumption per capita  $\dot{c}_j/c_j$ , respectively. Panel (d), (e), and (f) plots cross-country divergence in the level of technology  $A_j$ , income per capita  $y_j(t) = Y_j/L_j$ , and consumption per capita  $c_j = C_j/L_j$ , respectively. The solid red, dashed blue, and dotted black line corresponds to the USA, the EU, and RoW, respectively.

the long-run growth trend with all three countries/regions converging to a common growth rate of 5.52%. Technological interdependence provides the pulling force that ensures a similar growth rate across countries in the long-run sustained growth. This quantitative result coincides with the analytical finding of convergence obtained in Proposition 1.

**Divergence in Levels.** Although technological interdependence plays a role to generate cross-country convergence in the growth rates, the level of technology and income tends to diverge, owing to cross-country disparity in fundamental characteristics and the initial conditions of economic development. As Figure 4(d), (e), and (f) show, advanced economies, such as the USA and the EU, with higher indigenous innovation efficiencies and knowledge absorptive capacities, tend to generate higher levels of technology. Then, divergence in technology leads to cross-country differences in the level of income and consumption per capita. These results are in agreement with the analytical findings obtained in Proposition 3: cross-country disparities in structural characteristics lead to divergence in technology and income levels. Put differently, individual countries with different fundamental characteristics and initial conditions tend to grow along their country-specific balanced growth path, rather than converge toward a common growth path (e.g., the growth rate of the world technology frontier). Accordingly, our findings strengthen the viability of *club convergence* as a competing hypothesis with unconditional convergence: countries that differ in their structural characteristics and initial conditions may cluster around different steady-state/balanced growth equilibria [e.g., Galor (1996)].

**World Technology Distribution.** Starting from the initial condition of world technology distribution  $[a_{US}(0), a_{EU}(0), a_{RoW}(0)] = [0.29, 0.38, 0.33]$ , the USA and the EU will continue to have dominant shares in the world technology, while the contribution made by the RoW will decline. The long-run world technology distribution will end up with  $[a_{US}^*, a_{EU}^*, a_{RoW}^*] = [0.44, 0.33, 0.22]$ . This result suggests that technologically advanced economies with strong capacities to undertake indigenous innovation and absorb technology diffusion tend to account for larger shares in the long-run world technology distribution. In contrast, countries with relatively backward innovative capacity end up with a lower position in the world technology distribution.

**Comparative Statics.** Figure 5 shows the comparative static effects of technology characteristics on cross-country divergence. As shown in Figure 5(a), when the USA improves its indigenous innovation efficiency and knowledge absorptive capacity, the pace of technological progress tends to be faster in that country, widening the technology gap for both the EU and RoW relative to the USA. Meanwhile, Figure 5(b) shows that the faster technological progress in the USA also drives capital accumulation at a relatively larger rate as compared to that in both the EU and RoW. Hence, the capital ratio of the EU and RoW relative to the USA becomes smaller as a result of an increase in US indigenous innovation efficiency and knowledge absorptive capacity. Similar comparative statics results are reached for consumption and income as shown in Figure 5(c)–(f). The divergence between the EU/RoW and the USA tends to widen when the USA increases its indigenous innovation efficiency and knowledge absorptive capacity.



**FIGURE 5.** Panel (a) plots the comparative statics for technology ratios between the USA, the EU, and RoW. Panel (b) plots the comparative statics for capital ratios between the USA, the EU, and RoW. Panel (c) plots the comparative statics for consumption ratios between the USA, the EU, and RoW. Panel (d) plots the comparative statics for output ratios between the USA, the EU, and RoW. Panel (e) plots the comparative statics for per capita consumption ratios between the USA, the EU, and RoW. Panel (e) plots the comparative statics for per capita consumption ratios between the USA, the EU, and RoW. Panel (f) plots the comparative statics for per capita consumption ratios between the USA, the EU, and RoW. Panel (f) plots the comparative static for per capita income ratios between the USA, the EU, and RoW. The comparative static effects of structural characteristics are examined by considering 10% increases in the parameters of the US indigenous innovation efficiency and knowledge absorptive capacity from the benchmark values  $\lambda_{US} = 0.03$  and  $\sigma_{US} = 0.02$ .

#### 6. EMPIRICAL TESTS

#### 6.1. Empirical Strategy

This section describes empirical tests for the theoretical results obtained in previous sections. In our empirical analysis, we adopt intangible capital stock as a proxy indicator of technology. The reasons for this treatment are twofold. First, our Ramsey growth framework has factored into physical capital, that is,  $K_j$  in the production function (2). It is important to include intangible capital and link it to  $A_j$  in our empirical analysis. This follows from the fact that the advanced economy such as the USA and the EU has shifted toward service- and technology-based industries, which has made intangible assets such as innovative products, brands, patents, software, customer relationships, databases, and distribution systems increasingly important.

Second, in our growth model with cross-country technological interdependence, the technology level  $A_i$  is a stock variable that is cumulative over time (as described by the law of motion for technology), and technology across economies can be aggregated to be the world technology, that is,  $A_{WTP}(t) = \sum_{i=1}^{J} A_j(t)$ . These features can be captured using intangible capital which is both accumulative over time and aggregatable across countries. In contrast, total factor productivity as used in traditional growth accounting literature to measure technology is neither accumulative over time nor aggregatable across countries and is thus not applicable to our model. The method of using intangible capital as a proxy of technology builds on the literature of intangible capital and growth accounting including, among others, Corrado et al. (2009), Corrado and Hulten (2010), Nakamura (2010), Corrado and Hulten (2014), Eisfeldt and Papanikolaou (2014), Peters and Taylor (2017), Chen (2018), and Kogan and Papanikolaou (2019). It is widely acknowledged in these studies that intangible capital is closely related to total factor productivity, and the growth rates of output per worker are found to increase at a similar rate as that of intangible capital.

Specifically, we test the following three results obtained in the theoretical expositions. First, there is cross-country convergence in the rate of technological progress. Second, a country with a stronger capacity to undertake indigenous innovation and absorb technology diffusion tends to have a larger share in the world technology pool (a higher level of technology relative to other countries). Third, a country's structural characteristics affect its relative technology level. To test these results formally, we estimate the following equations:

$$\Delta \ln A_j(t) = a_j - c_j \left( \ln A_j(t-1) - d \ln A_{WTP}(t-1) \right) + \theta_1 \Delta \ln A_j(t-1) + \theta_2 \Delta A_{WTP}(t-1) + \vartheta_j(t),$$
(30a)

$$\Delta A_j(t) = e_j - f_j \left( A_j(t-1) - d_j A_{WTP}(t-1) \right) + \theta_3 \Delta A_j(t-1) + \theta_4 \Delta A_{WTP}(t-1) + \epsilon_i(t),$$
(30b)

$$d_i = \gamma_0 + \gamma_1 H K_i + \gamma_2 I S Q_i + \delta_i, \qquad (30c)$$

where  $A_j(t)$  is country j's technology level in period t,  $A_{WTP}(t-1)$  is the world technology pool in period (t-1),  $HK_j$  is country j's initial human capital stock, and  $ISQ_j$  is country j's initial institutional quality. Equation (30a) characterizes a panel cointegration model for all countries in the sample. The long-run equilibrium relationship is  $\ln A_j(t-1) = d \ln A_{WTP}(t-1)$ , with the coefficient of cointegration denoted by d. As Proposition 1 indicates, technology converges to a common growth rate across countries in the long run, and this common growth rate is equal to the growth rate of the world technology pool. This result suggests that the cointegrating coefficient should be equal to one, d = 1, since d corresponds to the ratio of  $A_j(t)$  which is the rate of technological progress in the economy j and the rate of the progress of the world technology  $A_{WTP}(t)$ .

Furthermore, we use the level of technology  $A_j(t)$  in (30b), instead of the natural logarithm of the technology  $\ln A_j(t)$  in (30a). The long-run relationship is  $A_j(t-1) = d_j A_{WTP}(t-1)$ , where  $d_j$  denotes the coefficient of cointegration that is country-specific. In particular, the values of  $d_j$  represent the share that a country j accounts for in the world technology pool in the long run. The specification given in (30b) is used to test the theoretical results in the proposition : a country with a stronger capacity to undertake indigenous innovation and absorb foreign technology diffusion accounts for a larger share in the world technology distribution. Then, we could test whether the value of  $d_j$  is greater when the capacity of innovation and absorption is stronger. Finally, (30c) examines how the value of  $d_j$  is influenced by variables that reflect a country's structural characteristics in terms of the capacity of indigenous innovation and technology adoption. Specifically, we test for the effect of human capital  $HK_j$  and institutional quality  $ISQ_j$  on the value of  $d_j$ .

#### 6.2. Data

The data source for our empirical tests is the latest version of the PWT 9.1 which contains data for variables required in the estimation. PWT 9.1 categorizes all capital in an economy into four types and provides data for each of them in the sample. These four types are (1) residential and nonresidential structure; (2) machinery and non-transport equipment; (3) transport equipment, and (4) other assets including software, intellectual property products, and cultivated assets. According to the study of Corrado et al. (2009), the intangible capital consists of three main categories: computerized information (software), intellectual property (R&D), and economic competency (advertising, staff training, and organization capital). The category "other assets" in the PWT 9.1 reflects the intangible capital conceptualized by Corrado et al. (2009). Therefore, we use the values of the "other assets" as a proxy for the intangible capital stock, which is the proxy for the technology level. In general, intangible capital consists of the stock of immaterial resources that enter the production process and is important for the creation and improvement of products as well as production processes [Arrighetti et al. (2014)]. It has been identified in the existing literature that intangible capital plays

an increasingly important role in determining firm productivity and economic performance [Marrocu et al. (2012)].

We also collect data for factors that may potentially determine a country's indigenous innovation efficiency and knowledge absorptive capacity: human capital stock  $HK_i$  and institutional quality  $ISQ_i$ . We include these two variables based on the findings in Comin and Hobijn (2004) that the most important determinants of the speed at which a country adopts technologies are the country's human capital endowment, type of government, degree of openness to trade, and adoption of predecessor technologies. Since institutional quality covers a measure of trade openness, we do not include another measure of trade openness separately. When estimating (30c), we use the initial values of the variables in the year 1970 which is the earliest year with all the variables required in the estimation. The human capital index in the year 1970 is obtained from PWT 9.1 and is used as a proxy for an initial value of the human capital stock. Initial values of institutional quality are measured by the economic freedom index in 1970, obtained from Fraser Institute's Economic Freedom Database. The database reports a chain-linked summary index that is the average of five subindices: (1) size of government; (2) legal structure and property rights; (3) access to sound money; (4) freedom to trade internationally; and (5) regulation of credit, labor, and business.

#### 6.3. Estimation Methods

The estimation is conducted in STATA 15. We run panel cointegration regressions based on the specifications in (30a) and (30b), and the ordinary least squares estimation for (30c). The empirical results are presented in Tables 2–4.

As we have 180 countries over 45 years (1970–2014) in the sample, the panel is of a large-N and large-T structure. There are several approaches to estimate the regression equation for a heterogeneous dynamic panel with both large N and large T. At one extreme, a fixed-effect (FE) estimation approach could be used in which the time series data for each group are pooled and only the intercepts are allowed to differ across groups. However, if the slope coefficients are not identical, the FE approach produces inconsistent and potentially misleading results. At the other extreme, the model could be fitted separately for each group, and a simple arithmetic average of the coefficients could be calculated. This is the mean group (MG) estimator proposed by Pesaran and Smith (1995). With this estimator, the intercepts, slope coefficients, and error variances are all allowed to differ across groups. In between the two extremes, Pesaran et al. (1999) propose a pooled mean group (PMG) estimator that combines both pooling and averaging. This intermediate estimator allows the intercept, short-run coefficients, and error variances to differ across the groups but constrains the long-run coefficients to be equal across groups.

We adopt the PMG estimator for (30a) and the MG estimator for (30b). The reason for using the two estimators above, respectively, is that for (30a) a common relationship is tested for all countries in the sample, that is, in the long

d	1.2*** (0.33)
Cointegration tests, under	Gt = -2.14 ( <i>p</i> -value=0.000) Ga = -10.42 ( <i>p</i> -value=0.000)
a null of no cointegration	Pt = -22.34 ( <i>p</i> -value=0.006) Pa = -5.86 ( <i>p</i> -value=0.00)
Number of observations	7113
Number of groups	180
Observations per group avg	40
Test $d = 1$	F = 1.81 (p-value=0.18)

 TABLE 2. The cointegration test of (30a)

Note. Standard errors in parenthesis. Significance levels are denoted as \*of 10%, \*\*of 5%, \*\*\*of 1%.

**TABLE 3.** The cointegration test of (30b)

All test statistics of panel cointegration	v = 7.36
are distributed $N(0, 1)$ under a null,	rho = -1.654
of no cointegration and diverge to	t = 2.68
negative infinity.	adf = -2.125

run, whether the growth rate of an individual country's technology level is equal to that of the world technology pool. Therefore, in the long-run relationship, the coefficient *d* is common across countries, and the suitable estimator is the PMG estimator. In contrast, (30b) identifies a pairwise relationship between each country and the world in terms of relative technology levels. In the long run, each country ends up with different technology levels relative to world technology. Therefore, the coefficient  $d_j$  varies across countries, and the more suitable estimator is the MG estimator.

#### 6.4. Results

In Tables 2 and 3, we can see that the test statistics for the cointegration test with the null of no cointegration are all statistically significant, thus rejecting the null of no cointegration. This finding suggests that there exist long-run relationships as specified in (30a)–(30b). In Table 4, the F-statistic is 1.81 and the associated *p*-value is 0.18. The test fails to reject the null at conventional significance levels that *d* is equal to 1 and therefore supports the result that growth rates of technology level are equal across economies in the long run.

In Table 4, we regress the values  $d_j$  of individual countries on a country's structural characteristics variables. The results suggest that both human capital and institutional quality are important factors determining a country's relative technology level in the long run. In other words, human capital is an essential input required for indigenous innovation and technology absorption, and institutional quality is also critical in shaping regulatory policy frameworks that enhance

Dependent variables	$d_j$
HK <sub>i</sub>	0.44**(0.20)
$ISQ_i$	0.23**(0.11)
Number of observations	53
R-squared	0.32

**TABLE 4.** The effect of explanatory variables on  $d_i$ , estimation of (30c)

*Note.* Standard errors in parenthesis. Significance levels are denoted as \*of 10%, \*\*of 5%, \*\*\*of 1%.

both domestic innovation and the absorption of knowledge spillovers from the RoW. These empirical results support the analytical findings in our model and are consistent with the theoretical works on the determinants of technology adoption [e.g., Parente and Prescott (1994), Comin and Hobijn (2004), Stokey (2015)].

It is worth noting that the sample size used to estimate (30c) is smaller than that used to estimate (30a) and (30b) because of limited data available for measuring institutional quality. Another caveat is that we are estimating the long-run relationship using an error correction specification in (30c) for the sample from 1970 to 2014. Although the sample period is long, it is possible that the economies are still in the phase of transitional dynamics and have not entered the BGP yet. Therefore, when interpreting the estimation results, we should bear this caveat in mind.

#### 7. CONCLUSION

The key takeaways from our analysis and findings are as follows. The world equilibrium with technological interdependence is neither simply the equilibrium of each country on its own nor the equilibrium resulting from unidirectional technology diffusion from a single frontier country that grows autonomously. Different countries grow at different rates during transitional dynamics but converge toward a common growth rate in the long run. The pulling force of technological interdependence ensures convergence in the long-run growth rates. The common growth rate for convergence is endogenously determined by structural characteristics of all individual countries in the world economy, rather than by a single frontier country that grows autonomously. Cross-country disparities in structural characteristics and initial conditions lead to divergence in technology and income levels, though technological interdependence ensures convergence in the growth rate. This finding rationalizes *club convergence*: the income levels tend to converge among countries that are identical in structural characteristics and initial conditions but diverge between countries with quite different characteristics and initial conditions.

There are still important caveats. First, our model focuses on specifications of technological interdependence, and the structural characteristics such as indigenous innovation efficiencies and knowledge absorptive capacities are described in

exogenous, parsimonious representations. One important direction for extensions is to provide endogenous specifications of indigenous innovation efficiency and knowledge absorptive capacity and to relate them to deeper determinants, such as R&D learning [e.g., Cohen and Levinthal (1989)], intellectual property rights [e.g., Marchese et al. (2019), Iwaisako (2020)], trade barriers [e.g., Parente and Prescott (1994)], education and human capital [e.g., Nelson and Phelps (1966), Stokey (2015), Chu et al. (2019), Dutt and Veneziani (2020)], labor market regulations [e.g., Alesina et al. (2018)], appropriate technology [e.g., Basu and Weil (1998), Acemoglu and Zilibotti (2001)], financial development [e.g., Trew (2014), Sunaga (2019)], and institutional arrangements [e.g., Acemoglu et al. (2006, 2017), Li et al. (2020)]. Second, there can potentially be cross-country linkages through the channel of international capital flows, besides the channel of technological spillovers and interdependence as emphasized in this paper. We expect that there might still be a cross-country pattern of convergence in the growth rates when international capital flows are introduced. But cross-country capital flows provide a channel for widening divergence in levels. Countries with a smaller rate of time preference (a high degree of patience) tend to maintain a positive asset position and consume the larger share of net world outputs, while countries with a larger rate of time preference (a low degree of patience) run a persistent current account deficit and consume less at later dates. The established results of both convergence in growth rates and divergence in levels might be robust. We leave details expositions of these areas for future research.

#### NOTES

1. The growth accounting literature shows that cross-country differences in income cannot be solely explained by differences in the physical or human capital, and technology differences across countries are likely to be at the heart of cross-country income differences [e.g., Barro (1991), Mankiw et al. (1992), Galor and Tsiddon (1997), Hall and Jones (1999), Galor (2005), Madsen (2008), Jones (2016)].

2. Cross-country differences in productivity and income have increased dramatically over the last 200 years. The average income ratio between Maddisons Western countries and the non-Western countries was 1.9 in 1820 and increased to 7.2 by 2000 [e.g., Maddison (2003)].

3. Without taking technological interdependence into account, the world behaves as an integrated economy rather than an interdependent collection of economies. There is no interaction between countries, and they simply grow in tandem. It would have sufficed to look at one economy only and analyze its growth compared with that of the world leader. Hence, the growth model would become a single-country one, and the equilibrium would be specific to each country on its own, rather than an interdependent world equilibrium.

4. We also acknowledge a strand of literature that examines growth and convergence through the channel of international trade such as Krugman (1979), Ventura (1997), Acemoglu and Ventura (2002), and Prettner and Strulik (2016).

5. For analytical simplicity, the focus of our analysis is on cross-country linkages through the channel of technological spillovers and interdependence. Cross-country interactions through the channel of international capital flows are omitted.

6. While indigenous innovation plays a critical role in improving technical efficiency, an additional source is to be found in the adoption of technologies already developed in other countries according

to macroeconomic evidences [e.g., Coe and Helpman (1995), Coe et al. (1997), Eaton and Kortum (1999)].

7. European Commission (2013) shows the trend of complementarity in the specialization patterns of three major world areas in the decade 2000–2010. The European Research Area is characterized by a marked specialization in transport and mechanical-related technologies. Asia is markedly specialized in Information and communications technology (ICT) and nanotechnology, and the USA presents a profile of strong specialization in pharmaceuticals, biotechnology, space, and the more service-oriented segments of ICT.

8. Dividing both sides of (6) by  $A_j(t)$  obtains  $\dot{A}_j(t)/A_j(t) = \lambda_j + \sigma_j \left( A_{WTP}(t)/A_j(t) - 1 \right) = \lambda_j + \sigma_j \left( a_j(t)^{-1} - 1 \right)$ , and substituting it into  $g_{WTP}(t) \coloneqq \dot{A}_{WTP}(t)/A_{WTP}(t) = \sum_{j=1}^J (A_j(t)/A_{WTP}(t))(\dot{A}_j(t)/A_{jj}(t)) = \sum_{j=1}^J a_j(t)\dot{A}_j(t)/A_j(t)$  derives (9).

9. In the long-run phase of balanced growth, variables with effective levels  $[a_i^*, k_j^*, barc_j^*]$  are stationary, and variables with aggregate levels  $[c_j(t), b_j(t), K_j(t), L_j(t), A_j(t)]$  increase in a balanced growth pattern.

10. As we tend to put our focus on cross-country disparity in technological characteristics such as indigenous innovation efficiencies and knowledge absorptive capacities, the rate of time preference and the coefficient of relative risk aversion are set at the same level across the three countries/regions.

11. Intangible capital consists of the stock of immaterial resources that enter the production process and are important for the creation or improvement of products as well as production processes [e.g., Arrighetti et al. (2014)]. The intangible capital is playing an important role in the productivity and performance of the economy [e.g., Marrocu et al. (2012)]. The intangible capital consists of three main categories: computerized information, intellectual property, and economic competency such as advertising, staff training, and organization capital [e.g., Corrado et al. (2009)].

#### REFERENCES

- Abramovitz, M. (1986) Catching up, forging ahead, and falling behind. *Journal of Economic History* 46(2), 385–406.
- Acemoglu, D. (2009) *Introduction to Modern Economic Growth*. Princeton and Oxford: Princeton University Press.
- Acemoglu, D., P. Aghion, L. Bursztyn and D. Hemous (2012) The environment and directed technical change. *American Economic Review* 102(1), 131–166.
- Acemoglu, D., P. Aghion and F. Zilibotti (2006) Distance to frontier, selection, and economic growth. *Journal of the European Economic Association* 4, 37–74.
- Acemoglu, D., J. Robinson and T. Verdier (2017) Asymmetric growth and institutions in an interdependent world. *Journal of Political Economy* 125(5), 1245–1305.
- Acemoglu, D. and J. Ventura (2002) The world income distribution. *Quarterly Journal of Economics* 117(2), 659–694.
- Acemoglu, D. and F. Zilibotti (2001) Productivity differences. *Quarterly Journal of Economics* 116(2), 563–606.
- Aghion, P. and P. Howitt (1998) Endogenous Growth Theory. Cambridge, MA: MIT Press.
- Alesina, A., M. Battisti and J. Zeira (2018) Technology and labor regulations: Theory and evidence. *Journal of Economic Growth* 23, 41–78.
- Arrighetti, A., F. Landini and A. Lasagni (2014) Intangible assets and firm heterogeneity: Evidence from Italy. *Research Policy* 43, 202–213.
- Azariadis, C. (1996) The economics of development traps. Journal of Economic Growth 1, 449-486.
- Barro, R. (1991) Economic growth in a cross section of countries. *Quarterly Journal of Economics* 106(2), 407–443.
- Barro, R. (2015) Convergence and modernisation. Economic Journal 125, 911-942.
- Barro, R. and X. Sala-i-Martin (1992) Convergence. Journal of Political Economy 100(2), 223–251.
- Barro, R. and X. Sala-i-Martin (1997) Technological diffusion, convergence, and growth. *Journal of Economic Growth* 2, 1–27.

- Basu, S. and D. Weil (1998) Appropriate technology and growth. *Quarterly Journal of Economics* 113(4), 1025–1054.
- Baumol, W. (1986) Productivity growth, convergence, and welfare: What the long-run data show. *American Economic Review* 76(5), 1072–1085.
- Benhabib, J., J. Perla and C. Tonetti (2014) Catch-up and fall-back through innovation and imitation. *Journal of Economic Growth* 19(3), 1–35.
- Bernard, A. B. and S. N. Durlauf (1995) Convergence in international output. Journal of Applied Econometrics 10(2), 97–108.
- Bernard, A. B. and S. N. Durlauf (1996) Interpreting tests of the convergence hypothesis. *Journal of Econometrics* 71(585), 161–173.
- Caselli, F. and W. J. Coleman (2001) Cross-country technology diffusion: The case of computers. *American Economic Review* 91(2), 328–335.
- Caselli, F., G. Esquivel and F. Lefort (1996) Reopening the convergence debate: A new look at crosscountry growth empirics. *Journal of Economic Growth* 40, 363–389.
- Chen, W. (2018) Cross-country income differences revisited: Accounting for the role of intangible capital. *Review of Income and Wealth* 64(3), 626–648.
- Chu, A., G. Cozzi and S. Galli (2014) Stage-dependent intellectual property rights. *Journal of Development Economics* 106, 239–249.
- Coe, D. and E. Helpman (1995) International R&D spillovers. *European Economic Review* 39(5), 859–887.
- Coe, D., E. Helpman and A. Hoffmaister (1997) North-south R&D spillovers. *Economic Journal* 107(5), 134–149.
- Cohen, W. and D. Levinthal (1989) Innovation and learning: The two faces of R&D. *Economic Journal* 99, 569–596.
- Comin, D. and B. Hobijn (2004) Cross-country technology adoption: Making the theories face the facts. *Journal of Monetary Economics* 51(1), 39–83.
- Comin, D. and B. Hobijn (2010) An exploration of technology diffusion. *American Economic Review* 100(5), 2031–2059.
- Comin, D. and M. Mestieri (2018) If technology has arrived everywhere, why has income diverged? *American Economic Journal: Macroeconomics* 10(3), 137–178.
- Corrado, C. and C. Hulten (2014) Innovation accounting. In: D. Jorgenson, S. Landefeld and P. Schreyer (eds.), *Measuring Economic Sustainability and Progress: Studies in Income and Wealth*, pp. 595–628. Chicago: University of Chicago Press.
- Corrado, C., C. Hulten and D. Sichel (2009) Intangible capital and U.S. economic growth. *Review of Income and Wealth* 55(3), 661–685.
- Corrado, C. and C. Hulten (2010) How do you measure a "technological revolution"? *American Economic Review* 100(2), 99–104.
- Eaton, J. and S. Kortum (1996) Trade in ideas patenting and productivity in the OECD. Journal of International Economics 40(3), 251–276.
- Eaton, J. and S. Kortum (1999) International technology diffusion: Theory and measurement. *International Economic Review* 40(3), 537–570.
- Eaton, J. and S. Kortum (2001) Technology, trade, and growth: A unified framework. *European Economic Review* 45(4–6), 742–755.
- Eisfeldt, A. and D. Papanikolaou (2014) The value and ownership of intangible capital. *American Economic Review* 104(5), 189–194.
- Epstein, L. and S. Zin (1991) Substitution, risk aversion, and the temporal behavior of consumption and asset returns: An empirical analysis. *Journal of Political Economy* 99(2), 263–286.
- European Commission (2013) *The Technological Profile and Specialization Pattern of Countries*. Luxembourg: Publications Office of the European Union.
- Feenstra, R., R. Inklaar and M. Timmer (2015) The next generation of the Penn World Table. American Economic Review 105(10), 3150–3182.
- Galor, O. (1996) Convergence? Inference from theoretical models. *Economic Journal* 106, 1056–1069.

#### 1368 WEI JIN AND YIXIAO ZHOU

- Galor, O. (2005) From stagnation to growth: Unified growth theory. In: P. Aghion and S. Durlauf (eds.), *Handbook of Economic Growth*, pp. 171–293. Amsterdam: North-Holland.
- Galor, O. and D. Tsiddon (1997) Tecnological progress, mobility, and growth. American Economic Review 87(3), 363–382.
- Gerschenkron, A. (1962) *Economic Backwardness in Political Perspective*. Chicago: University of Chicago Press.
- Grossmann, V., T. Steger and T. Trimborn (2013) Dynamically optimal R&D subsidization. Journal of Economic Dynamics and Control 37(3), 516–534.
- Hall, R. and C. Jones (1999) Why do some countries produce so much more output per worker than others? *Quarterly Journal of Economics* 114(1), 83–116.
- Howitt, P. (2000) Endogenous growth and cross-country income differences. *American Economic Review* 90(4), 829–846.
- Howitt, P. and D. Mayer-Foulkes (2005) R&D, implementation and stagnation: A Schumpeterian theory of convergence clubs. *Journal of Money, Credit and Banking* 37(1), 147–177.
- Jones, C. (1995) R&D-based models of economic growth. *Journal of Political Economy* 103(4), 759–784.
- Jones, C. (1999) Growth: With or without scale effects?. American Economic Review 89(2), 139-144.
- Jones, C. (2016) The facts of economic growth. In: J. Taylor and H. Uhlig (eds.), Handbook of Macroeconomics, pp. 3–69. Elsevier.
- Kogan, L. and D. Papanikolaou (2019) Technological innovation, intangible capital, and asset prices. Annual Review of Financial Economics 11(1), 221–242.
- Krugman, P. (1979) A model of innovation, technology transfer, and the world distribution of income. *Journal of Political Economy* 87(2), 253–266.
- Maddison, A. (2003) *The World Economy: Historical Statistics*. Paris: Organisation for Economic Cooperation and Development.
- Madsen, J. (2008) Semi-endogenous versus Schumpeterian growth models: Testing the knowledge production function using international data. *Journal of Economic Growth* 13, 1–26.
- Mankiw, G., D. Romer and D. Weil (1992) A contribution to the empirics of economic growth. *The Quarterly Journal of Economics* 107(2), 407–437.
- Marrocu, E., R. Paci and M. Pontis (2012) Intangible capital and firms' productivity. *Industrial and Corporate Change* 21(2), 377–402.
- Mehra, R. and E. Prescott (1985) The equity premium: A puzzle. *Journal of Monetary Economics* 15(2), 145–161.
- Nakamura, L. (2010) Intangible assets and national income accounting. *Review of Income and Wealth* 56, S135–S155.
- Nelson, R. and E. Phelps (1966) Investment in humans, technological diffusion, and economic growth. *American Economic Review* 56(1), 69–75.
- OECD (2014) How was Life? Global Well-being since 1820. Paris: OECD Publishing.
- Parente, S. and E. Prescott (1994) Barriers to technology adoption and development. *Journal of Political Economy* 102(2), 298–321.
- Pesaran, H., Y. Shin and R. Smith (1999) Pooled mean group estimation of dynamic heterogeneous panels. *Journal of the American Statistical Association* 94(446), 621–634.
- Pesaran, H. and R. Smith (1995) Estimating long-run relationships from dynamic heterogeneous panels. *Journal of Econometrics* 68(1), 79–113.
- Peters, R. and L. Taylor (2017) Intangible capital and the investment-q relation. Journal of Financial Economics 123(2), 251–272.
- Quah, D. (1997) Empirics for growth and distribution: Stratication, polarization, and convergence clubs. *Journal of Economic Growth* 2, 27–59.
- Quah, D. (1996) Twin peaks: Growth and convergence in models of distribution dynamics. *Economic Journal* 106, 1045–1055.
- Romer, P. (1990) Endogenous technological change. Journal of Political Economy 98(5), S71–S102.
- Rosenberg, N. (1982) Inside the Black Box: Technology and Econonmics. New York: Cambridge University Press.

Stokey, N. (2015) Catching up and falling behind. Journal of Economic Growth 20, 1-36.

- van der Ploeg, F. and C. Withagen (2014) Growth, renewables, and the optimal carbon tax. *International Economic Review* 55, 283–311.
- Ventura, J. (1997) Growth and interdependence. Quarterly Journal of Economics 112(1), 57-84.
- Williamson, J. (1996) Globalization, convergence, and history. *Journal of Economic History* 56(2), 277–306.

World Bank (2017) World Development Indicators 2017. Washington, D.C.: World Bank.

Zeira, J. (1998) Workers, machines, and economic growth. *Quarterly Journal of Economics* 113(4), 1091–1117.

# APPENDIX A: TRANSITIONAL DYNAMICS STABILITY

Local stability of transitional dynamics is investigated by evaluating the Jacobian matrix of (10), (11) and (12) at the steady state  $[a_i^*, k_i^*, \bar{c}_i^*]$ :

$$\begin{bmatrix} \frac{\partial \dot{k}_{j}}{\partial k_{j}} & \frac{\partial \dot{k}_{j}}{\partial a_{j}} & \frac{\partial \dot{k}_{j}}{\partial \bar{c}_{j}} \\ \frac{\partial \dot{a}_{j}}{\partial k_{j}} & \frac{\partial \dot{a}_{j}}{\partial a_{j}} & \frac{\partial \dot{a}_{j}}{\partial \bar{c}_{j}} \\ \frac{\partial \dot{c}_{j}}{\partial k_{j}} & \frac{\partial \dot{c}_{j}}{\partial a_{j}} & \frac{\partial \dot{c}_{j}}{\partial \bar{c}_{j}} \end{bmatrix}$$

$$= \begin{bmatrix} f'(k_{j}^{*}) - \left[n_{j} + \delta_{j} + \lambda_{j} + \sigma_{j}\left(\frac{1}{a_{j}^{*}} - 1\right)\right] & \frac{\sigma_{j}k_{j}^{*}}{(a_{j}^{*})^{2}} & -1 \\ 0 & -g_{WTP}^{*} - (\lambda_{j} - \sigma_{j})(a_{j}^{*} - 1) & 0 \\ \frac{1}{\theta_{j}}\bar{c}_{j}^{*}f''(k_{j}^{*}) & \frac{\sigma_{j}\bar{c}_{j}^{*}}{(a_{j}^{*})^{2}} & 0 \end{bmatrix}.$$
(A1)

The determinant of the Jacobian matrix (A1) boils down to a three-order polynominal equation as follows:

$$\begin{pmatrix} \chi_j + g_{WTP}^* + (\lambda_j - \sigma_j)(a_j^* - 1) \end{pmatrix} \\ \times \left( \chi_j^2 - \left( f'(k_j^*) - n_j - \delta_j - \lambda_j - \sigma_j \left( \frac{1}{a_j^*} - 1 \right) \right) \chi_j + \frac{\bar{c}_j^* f''(k_j^*)}{\theta_j} \right) = 0,$$
 (A2)

where the root of the polynominal equation,  $\chi_j$ , corresponds to the eigenvalue of the Jacobian matrix (A1). The three-order polynominal equation has three roots, and one of the three eigenvalues takes the form  $\chi_j = -g^*_{WTP} - (\lambda_j - \sigma_j)(a^*_j - 1) = -\left(\frac{\sigma_j}{a^*_j} + (\lambda_j - \sigma_j)a^*_j\right) < 0$ , where the second equality follows from the first expression of (13), that is,  $g^*_{WTP} = \lambda_j - \sigma_j + \sigma_j/a^*_j$ .  $\lambda_j > \sigma_j$  pins down the negative sign. Meanwhile, given that  $\bar{c}^*_j f''(k^*_j)/\theta_j < 0$  in (A2), the other two eigenvalues are one positive and one negative. The negative eigenvalues of the Jacobian matrix thus establish the saddle-path stability of transitional dynamics.

## APPENDIX B: PROOF OF COROLLARY 1

To prove  $\partial g_{WTP}^*/\partial \lambda_j > 0$ , differentiating (14) with respect to  $\lambda_j$  yields

$$rac{\sigma_j}{\left(\sigma_j-\lambda_j+g^*_{WTP}
ight)^2}rac{\partial\left(\sigma_j-\lambda_j+g^*_{WTP}
ight)}{\partial\lambda_j}+\sum_{i
eq j}rac{\sigma_i}{\left(\sigma_i-\lambda_i+g^*_{WTP}
ight)^2}rac{\partial\left(\sigma_i-\lambda_i+g^*_{WTP}
ight)}{\partial\lambda_j}=0,$$

and simplifying the terms yields (16a). Similarly, differentiating (14) with respect to  $\sigma_i$  yields

$$\frac{1}{\sigma_j - \lambda_j + g^*_{WTP}} - \frac{\sigma_j}{\left(\sigma_j - \lambda_j + g^*_{WTP}\right)^2} \frac{\partial \left(\sigma_j - \lambda_j + g^*_{WTP}\right)}{\partial \sigma_j} - \sum_{i \neq j} \frac{\sigma_i}{\left(\sigma_i - \lambda_i + g^*_{WTP}\right)^2} \frac{\partial \left(\sigma_i - \lambda_i + g^*_{WTP}\right)}{\partial \sigma_j} = 0,$$

and simplifying the terms yields (16b), where the positive sign follows from  $g_{WTP}^* - \lambda_j > 0$  given by (14).

# **APPENDIX C: PROOF OF PROPOSITION 2**

Differentiating the first expression of (13) with respect to  $\lambda_i$  yields

$$\frac{\partial a_j^*}{\partial \lambda_j} = \left(\frac{\sigma_j - \lambda_j + g_{WTP}^*}{\sigma_j}\right)^{-2} \left(\frac{1}{\sigma_j} - \frac{1}{\sigma_j} \frac{\partial g_{WTP}^*}{\partial \lambda_j}\right) \\ = \left(\frac{\sigma_j - \lambda_j + g_{WTP}^*}{\sigma_j}\right)^{-2} \left[\frac{\left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2 \sum_{i=1}^J \frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2} - \sigma_j}{\sigma_j \left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2 \sum_{i=1}^J \frac{\sigma_i}{\left(\sigma_i - \lambda_i + g_{WTP}^*\right)^2}}\right],$$
(C1)

where (16a) is used to replace  $\partial g^*_{WTP} / \partial \lambda_j$  in the first equality of (C1), and the nominator in the last term on the right-hand side of (C1) can be simplified as:

$$\begin{bmatrix} \left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2} \sum_{i=1}^{J} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}} \end{bmatrix} - \sigma_{j}$$
$$= \left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2} \sum_{i \neq j} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}}.$$
 (C2)

We hence obtain (17a). Similarly, differentiating the first expression of (13) with respect to  $\sigma_j$  yields

$$\frac{\partial a_{j}^{*}}{\partial \sigma_{j}} = \left(\frac{\sigma_{j} - \lambda_{j} + g_{WTP}^{*}}{\sigma_{j}}\right)^{-2} \left(\frac{g_{WTP}^{*} - \lambda_{j}}{\sigma_{j}^{2}} - \frac{1}{\sigma_{j}} \frac{\partial g_{WTP}^{*}}{\partial \sigma_{j}}\right)$$
$$= \left(\frac{\sigma_{j} - \lambda_{j} + g_{WTP}^{*}}{\sigma_{j}}\right)^{-2} \left(\frac{g_{WTP}^{*} - \lambda_{j}}{\sigma_{j}}\right) \left[\frac{\left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2} \sum_{j=1}^{J} \frac{\sigma_{j}}{\left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2}} - \sigma_{j}}{\sigma_{j}\left(\sigma_{j} - \lambda_{j} + g_{WTP}^{*}\right)^{2} \sum_{i=1}^{J} \frac{\sigma_{i}}{\left(\sigma_{i} - \lambda_{i} + g_{WTP}^{*}\right)^{2}}\right].$$
(C3)

Using (C2) to simplify the terms of (C3) yields (17b).

We proceed by differentiating the first expression of (13) with respect to  $\lambda_i$  yields

$$\begin{split} \frac{\partial a_{i}^{*}}{\partial \lambda_{j}} &= -\left(\frac{\sigma_{i} - \lambda_{i} + g_{WTP}^{*}}{\sigma_{i}}\right)^{-2} \left(\frac{1}{\sigma_{i}} \frac{\partial g_{WTP}^{*}}{\partial \lambda_{j}}\right) < 0,\\ \frac{\partial a_{i}^{*}}{\partial \sigma_{j}} &= -\left(\frac{\sigma_{i} - \lambda_{i} + g_{WTP}^{*}}{\sigma_{i}}\right)^{-2} \left(\frac{1}{\sigma_{i}} \frac{\partial g_{WTP}^{*}}{\partial \sigma_{j}}\right) < 0, \end{split}$$

and using (16a) to replace the expression of  $\frac{\partial g_{WTP}^*}{\partial \lambda_j}$  yields (17c) and (17d).

# APPENDIX D: PROOF OF PROPOSITION 3

First, imposing the stationary condition on  $\dot{a}_j/a_j = \dot{A}_j/A_j - \dot{A}_{WTP}/A_{WTP}$  yields  $\dot{A}_i^*/A_i^* = \dot{A}_j^*/A_j^* = \dot{A}_{WTP}^*/A_{WTP}^* := g_{WTP}^*$  for countries i, j = 1, 2, ..., J. Given  $\dot{A}_i^*/A_i^* = \dot{A}_j^*/A_j^* = \dot{A}_{WTP}^*/A_{WTP}^*$ , we have  $A_i^*/A_j^* = a_i^*/a_j^*$ . Then substituting the first expression of (13) yields (18). Second, following the second expression of (13), we derive

$$rac{k_i^*}{k_j^*} = rac{f'^{-1} imes \left(\delta_i + 
ho_i + heta_i g^*_{WTP}
ight)}{f'^{-1} \left(\delta_j + 
ho_j + heta_j g^*_{WTP}
ight)},$$

which yields (19). Finally, following the last expression of (13), we have

$$\begin{aligned} \frac{c_i^*}{c_j^*} &= \frac{A_i^*}{A_j^*} \frac{\bar{c}_i^*}{\bar{c}_j^*} = \left( \frac{\frac{\sigma_i}{(\sigma_i - \lambda_i + g_{WTP}^*)}}{\frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)}} \right) \\ &\times \left( \frac{f(f'^{-1}\left(\delta_i + \rho_i + \theta_i g_{WTP}^*\right)) - \left(\delta_i + n_i + g_{WTP}^*\right) f'^{-1}\left(\delta_i + \rho_i + \theta_i g_{WTP}^*\right)}{f(f'^{-1}\left(\delta_j + \rho_j + \theta_j g_{WTP}^*\right)) - \left(\delta_j + n_j + g_{WTP}^*\right) f'^{-1}\left(\delta_j + \rho_j + \theta_j g_{WTP}^*\right)} \right), \end{aligned}$$

which yields (20).

### APPENDIX E: PROOF OF PROPOSITION 4

Given that  $A_i^*/A_i^* = a_i^*/a_i^*$ , differentiating  $a_i^*/a_i^*$  with respect to  $\lambda_i$ ,  $\sigma_i$ ,  $\lambda_j$  and  $\sigma_j$  yields

$$\frac{\partial (a_i^*/a_j^*)}{\partial \lambda_i} = \frac{1}{a_j^*} \frac{\partial a_i^*}{\partial \lambda_i} - \frac{a_i^*}{(a_j^*)^2} \frac{\partial a_j^*}{\partial \lambda_i} > 0, \qquad \frac{\partial (a_i^*/a_j^*)}{\partial \sigma_i} = \frac{1}{a_j^*} \frac{\partial a_i^*}{\partial \sigma_i} - \frac{a_i^*}{(a_j^*)^2} \frac{\partial a_j^*}{\partial \sigma_i} > 0, \quad (\mathbf{E1a})$$

$$\frac{\partial(a_i^*/a_j^*)}{\partial\lambda_j} = \frac{1}{a_j^*} \frac{\partial a_i^*}{\partial\lambda_j} - \frac{a_i^*}{(a_j^*)^2} \frac{\partial a_j}{\partial\lambda_j} < 0, \qquad \frac{\partial(a_i^*/a_j^*)}{\partial\sigma_j} = \frac{1}{a_j^*} \frac{\partial a_i^*}{\partial\sigma_j} - \frac{a_i^*}{(a_j^*)^2} \frac{\partial a_j^*}{\partial\sigma_j} < 0, \quad (E1b)$$

where the signs of (E1a)–(E1b) are determined following  $\frac{\partial a_i^*}{\partial \lambda_i} > 0$ ,  $\frac{\partial a_j^*}{\partial \lambda_i} < 0$ ,  $\frac{\partial a_i^*}{\partial \sigma_i} > 0$ ,  $\frac{\partial a_j^*}{\partial \sigma_i} < 0$ ,  $\frac{\partial a_i^*}{\partial \sigma_i} < 0$ ,  $\frac{\partial a_i^*}{\partial \sigma_i} > 0$ ,  $\frac{\partial a_i^*}{\partial \sigma_i} > 0$  and  $\frac{\partial a_j^*}{\partial \sigma_j} > 0$  as given by (6.1).

For the within-country effect, differentiating the second and third expression of (13) with respect to  $\lambda_i$  and  $\sigma_i$  yields

$$\frac{\partial k_j^*}{\partial \lambda_j} = \frac{\theta_j}{f''(k_j^*)} \frac{\partial g_{WTP}^*}{\partial \lambda_j} < 0,$$
(E2a)

$$\frac{\partial k_j^*}{\partial \sigma_j} = \frac{\theta_j}{f''(k_j^*)} \frac{\partial g_{WTP}^*}{\partial \sigma_j} < 0,$$
(E2b)

$$\frac{\partial \bar{c}_j^*}{\partial \lambda_j} = \left( \left( (\theta_j - 1) g_{WTP}^* + \rho_j - n_j \right) \frac{\theta_j}{f''(k_j^*)} - k_j^* \right) \frac{\partial g_{WTP}^*}{\partial \lambda_j} < 0,$$
(E2c)

$$\frac{\partial \bar{c}_j^*}{\partial \sigma_j} = \left( \left( (\theta_j - 1)g_{WTP}^* + \rho_j - n_j \right) \frac{\theta_j}{f''(k_j^*)} - k_j^* \right) \frac{\partial g_{WTP}^*}{\partial \sigma_j} < 0,$$
(E2d)

where  $(\theta_j - 1)g_{WTP}^* - n_j > 0$  with  $\theta_j > 1$ , and the negative sign follows from  $f''(k_j^*) < 0$ ,  $\partial g_{WTP}^*/\partial \lambda_j > 0$ , and  $\partial g_{WTP}^*/\partial \sigma_j > 0$ . Meanwhile, for the cross-country effect, differentiating the second and third expression of (13) with respect to  $\sigma_j$  and  $\lambda_j$  yields

$$\frac{\partial k_i^*}{\partial \lambda_j} = \frac{\theta_i}{f''(k_i^*)} \frac{\partial g_{WTP}^*}{\partial \lambda_j} < 0,$$
(E3a)

$$\frac{\partial k_i^*}{\partial \sigma_j} = \frac{\theta_i}{f''(k_i^*)} \frac{\partial g_{WTP}^*}{\partial \sigma_j} < 0,$$
(E3b)

$$\frac{\partial \bar{c}_i^*}{\partial \lambda_j} = \left[ \left( (\theta_i - 1) g_{WTP}^* + \rho_i - n_i \right) \frac{\theta_i}{f''(k_i^*)} - k_i^* \right] \frac{\partial g_{WTP}^*}{\partial \lambda_j} < 0,$$
(E3c)

$$\frac{\partial \bar{c}_i^*}{\partial \sigma_j} = \left[ \left( (\theta_i - 1) g_{WTP}^* + \rho_i - n_i \right) \frac{\theta_i}{f''(k_i^*)} - k_i^* \right] \frac{\partial g_{WTP}^*}{\partial \sigma_j} < 0,$$
(E3d)

where the negative sign follows from  $(\theta_i - 1)g^*_{WTP} + \rho_i - n_i > 0$ ,  $f''(k^*_i) < 0$ ,  $\frac{\partial g^*_{WTP}}{\partial \lambda_j} > 0$  and  $\frac{\partial g^*_{WTP}}{\partial \sigma_j} > 0$ .

For the comparative static effect on  $\frac{K_i^*}{K_j^*} = \frac{k_i^*}{k_j^*} \frac{A_i^*}{A_j^*} \frac{L_i^*}{L_j^*} = \frac{k_i^*}{k_j^*} \frac{a_i^*}{a_j^*} \frac{L_i^*}{L_j^*}$ , differentiating  $\frac{K_i^*}{K_j^*}$  with respect to  $\lambda_i$  and  $\sigma_i$  yields

$$\frac{\partial (K_i^*/K_j^*)}{\partial \lambda_i} = \left(\frac{a_i^*}{a_j^*} \cdot \frac{\partial (k_i^*/k_j^*)}{\partial \lambda_i} + \frac{k_i^*}{k_j^*} \cdot \frac{\partial (a_i^*/a_j^*)}{\partial \lambda_i}\right) \frac{L_i^*}{L_j^*},\tag{E4}$$

where the first term in the right-hand side of (E4) is given by:

$$\frac{a_i^*}{a_j^*} \cdot \frac{\partial(k_i^*/k_j^*)}{\partial \lambda_i} = \frac{a_i^*}{a_j^*} \cdot \left(\frac{1}{k_j^*} \frac{\partial k_i^*}{\partial \lambda_i} - \frac{k_i^*}{(k_j^*)^2} \frac{\partial k_j^*}{\partial \lambda_i}\right) = \frac{a_i^*}{a_j^*} \frac{k_i^*}{k_j^*} \left(\frac{\theta_i}{k_i^* f''(k_i^*)} - \frac{\theta_j}{k_j^* f''(k_j^*)}\right) \frac{\partial g_{WTP}^*}{\partial \lambda_i},$$

and the second term in the right-hand side of (E4) is given by:

$$\frac{k_i^*}{k_j^*} \cdot \frac{\partial(a_i^*/a_j^*)}{\partial \lambda_i} = \frac{k_i^*}{k_j^*} \cdot \left(\frac{1}{a_j^*} \frac{\partial a_i^*}{\partial \lambda_i} - \frac{a_i^*}{(a_j^*)^2} \frac{\partial a_j^*}{\partial \lambda_i}\right)$$
$$= \frac{a_i^*}{a_j^*} \frac{k_i^*}{k_j^*} \left(\frac{1}{a_i^*} \sum_{j \neq i} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} - \frac{1}{a_j^*} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2}\right) \frac{\partial g_{WTP}^*}{\partial \lambda_i}.$$

We thus simplify (E4) as:

$$\begin{aligned} \frac{\partial (K_i^*/K_j^*)}{\partial \lambda_i} &= \frac{a_i^*}{a_j^*} \frac{k_i^*}{k_j^*} \frac{L_i^*}{L_j^*} \frac{\partial g_{WTP}^*}{\partial \lambda_i} \\ &\times \left( \frac{\theta_i}{k_i^* f''(k_i^*)} - \frac{\theta_j}{k_j^* f''(k_j^*)} + \frac{1}{a_i^*} \sum_{j \neq i} \frac{\sigma_j}{\left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2} - \frac{1}{a_j^*} \frac{\sigma_j}{\left(\sigma_j - \lambda_j + g_{WTP}^*\right)^2} \right), \end{aligned}$$

and obtain the condition (22). Similarly, we can show that  $\frac{\partial (K_i^*/K_j^*)}{\partial \sigma_i} > 0$ ,  $\frac{\partial (K_i^*/K_j^*)}{\partial \lambda_j} < 0$ , and  $\frac{\partial (K_i^*/K_j^*)}{\partial \sigma_j} < 0$  if the condition (22) holds.

For the comparative static effect on  $\frac{C_i^*}{C_j^*} = \frac{\tilde{c}_i^*}{\tilde{c}_j^*} \frac{A_i^*}{A_j^*} \frac{L_i^*}{L_j^*} = \frac{\tilde{c}_i^*}{\tilde{c}_j^*} \frac{a_i^*}{a_j^*} \frac{L_i^*}{L_j^*}$ , differentiating  $\frac{C_i^*}{C_j^*}$  with respect to  $\lambda_i$  and  $\sigma_i$  yields

$$\frac{\partial (C_i^*/C_j^*)}{\partial \lambda_i} = \left(\frac{a_i^*}{a_j^*} \cdot \frac{\partial (\bar{c}_i^*/\bar{c}_j^*)}{\partial \lambda_i} + \frac{\bar{c}_i^*}{\bar{c}_j^*} \cdot \frac{\partial (a_i^*/a_j^*)}{\partial \lambda_i}\right) \frac{L_i^*}{L_j^*},\tag{E5}$$

where the first term in the right-hand side of (E5) is given by:

$$\begin{split} \frac{a_i^*}{a_j^*} \cdot \frac{\partial(\bar{c}_i^*/\bar{c}_j^*)}{\partial \lambda_i} &= \frac{a_i^*}{a_j^*} \cdot \left(\frac{1}{\bar{c}_j^*} \frac{\partial \bar{c}_i^*}{\partial \lambda_i} - \frac{\bar{c}_i^*}{(\bar{c}_j^*)^2} \frac{\partial \bar{c}_j^*}{\partial \lambda_i}\right) \\ &= \frac{a_i^*}{a_j^*} \frac{\bar{c}_i^*}{\bar{c}_j^*} \left[\frac{1}{\bar{c}_i^*} \left(\left((\theta_i - 1)g_{WTP}^* + \rho_i - n_i\right) \frac{\theta_i}{f''(k_i^*)} - k_i^*\right) \right. \\ &\left. - \frac{1}{\bar{c}_j^*} \left(\left((\theta_j - 1)g_{WTP}^* + \rho_j - n_j\right) \frac{\theta_j}{f''(k_j^*)} - k_j^*\right)\right] \frac{\partial g_{WTP}^*}{\partial \lambda_i}, \end{split}$$

and the second term in the right-hand side of (E5) is given by:

$$\begin{split} \frac{\bar{c}_i^*}{\bar{c}_j^*} \cdot \frac{\partial(a_i^*/a_j^*)}{\partial \lambda_i} &= \frac{\bar{c}_i^*}{\bar{c}_j^*} \cdot \left( \frac{1}{a_j^*} \frac{\partial a_i^*}{\partial \lambda_i} - \frac{a_i^*}{(a_j^*)^2} \frac{\partial a_j^*}{\partial \lambda_i} \right) \\ &= \frac{a_i^*}{a_j^*} \frac{\bar{c}_i^*}{\bar{c}_j^*} \left( \frac{1}{a_i^*} \sum_{j \neq i} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} - \frac{1}{a_j^*} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} \right) \frac{\partial g_{WTP}^*}{\partial \lambda_i}, \end{split}$$

We thus simplify (E5) as:

$$\begin{aligned} \frac{\partial(C_i^*/C_j^*)}{\partial \lambda_i} &= \frac{a_i^*}{a_j^*} \frac{\bar{c}_i^*}{\bar{c}_j^*} \frac{L_i^*}{L_j^*} \frac{\partial g_{WTP}^*}{\partial \lambda_i} \left[ \frac{1}{a_i^*} \sum_{j \neq i} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} - \frac{1}{a_j^*} \frac{\sigma_j}{(\sigma_j - \lambda_j + g_{WTP}^*)^2} \right] \\ &+ \frac{1}{\bar{c}_i^*} \left( \left( (\theta_i - 1)g_{WTP}^* + \rho_i - n_i \right) \frac{\theta_i}{f''(k_i^*)} - k_i^* \right) \\ &- \frac{1}{\bar{c}_j^*} \left( \left( (\theta_j - 1)g_{WTP}^* + \rho_j - n_j \right) \frac{\theta_j}{f''(k_j^*)} - k_j^* \right) \end{aligned}$$

and obtain the condition (25). Similarly, we can show that  $\frac{\partial (C_i^*/C_j^*)}{\partial \sigma_i} > 0$ ,  $\frac{\partial (C_i^*/C_j^*)}{\partial \lambda_j} < 0$ , and  $\frac{\partial (C_i^*/C_j^*)}{\partial \sigma_i} < 0$  if the condition (25) holds.

# APPENDIX F: PROOF OF PROPOSITION 5

For the negative eigenvalue given by (29),  $\frac{\sigma_j}{a_j^*} + (\lambda_j - \sigma_j)a_j^*$  is decreasing in  $a_j^*$  when  $a_j^*$  is within the range  $a_j^* \in [0, \sqrt{\sigma_j(\lambda_j - \sigma_j)}/\sigma_j]$ . As the value of the technology gap decreases  $a_j^* \downarrow$ , the absolute value of the negative eigenvalue increases  $|\chi_j| \uparrow$ , thus translating into a larger speed of convergence and a shorter time of transitional dynamics.